THE NATIONAL PARLIAMENTS' APPROVAL ON THE EU MEMBER STATES' ARMS EXPORTS

LEIDEN UNIVERSITY

PANDU WIJAYA [S1203991]

Supervisor: Dr. Eugenio Cusumano MASTER THESIS EUROPEAN UNION STUDIES Leiden University 16 January 2015 Leiden, the Netherlands Number of words (content): 20.227



Acknowledgement

I make this thesis to encourage the improvement of the decision making process in the export of arms in the Netherlands and the improvement of the respect and protection on human rights in Indonesia. In addition, by making this thesis, I also expect that there will be an improvement in the bilateral relationship between the Netherlands and Indonesia.

This thesis is the sum of the contributions of many, who have inspired, encouraged, supported during my study following the European Union Studies in Leiden University. First and foremost, I thank Dr. Eugenio Cusumano who has helped me so much in finishing this thesis in a quite short time. I am also very much thankful for the support of Dr. Jan Oster giving me help and big encouragement during my graduation period. Without their support, I might have not been able to graduate in time.

My gratitude also goes to the lecturers and friends from the master program European Union Studies and from other master programs that I follow in Leiden University, and my friends in Indonesia and my friends in the Netherlands, who have encouraged and supported me during my study and who have made me feel better in my hard days. I am also thankful to my friend Addison Dodd for his willingness to be the second reader for this thesis.

Finally, my deepest appreciation goes to my beloved husband, Alexander Arends; my parents, my brother Wisnu and his wife, and other family members in Indonesia; my parents in law, my sisters in law and their spouses, and other family members of family Arends in the Netherlands, whose support has been very much indispensable to me during my study. Without them I would never be able to finish this thesis.

THE NATIONAL PARLIAMENTS' APPROVAL ON THE EU MEMBER STATES' ARMS EXPORTS	6
Abstract	6
CHAPTER 1 INTRODUCTION	7
THESIS AIM	7
How governments' attempts earn parliamentary approval	8
RESEARCH QUESTION	9
THESIS LIMITATIONS AND THE FOCUS OF THE STUDY	9
THEORETICAL FRAMEWORK	11
RESEARCH METHOD AND SOURCES	12
THESIS OUTLINE	13
CHAPTER 2 LITERATURE REVIEW	16
THE MOTIVES BEHIND THE ESTABLISHMENT OF THE ECC	16
THE EFFECTIVENESS OF THE ECC	16
THE IMPACTS OF THE OBLIGATION TO IMPLEMENT THE ECC'S CRITERIA IN THE ARMS EXPORT CONTROL MECHANISM IN THE NETHERLANDS	16
THE HISTORY OF THE DUTCH REGULATIONS FOR THE ARMS EXPORT CONTROLS IN THE NETHERLANDS	17
THE ARMS EXPORT CONTROLS IN PRACTICE IN THE NETHERLANDS	18
THE ROLE OF THE DUTCH MINISTRIES IN THE ARMS EXPORT CONTROLS IN THE NETHERLANDS	18
THE ROLE OF DUTCH PARLIAMENTS IN THE ARMS EXPORT CONTROLS IN THE NETHERLANDS	19
WHY PARLIAMENTS' APPROVAL IS IMPORTANT	20
THE SOURCES TO EARN PARLIAMENTS' APPROVAL	20
EXPLANATORY REASONS EXPLAINING WHY PARLIAMENTS IN OTHER MEMBER STATES APPROVE THE ATTEMPTS OF THEIR GOVERNMENTS TO GRANT THE PERMITS TO ARMS EXPORTS TO THIRD COUNTRIES CONSIDERED AS HUMAN RIGHTS PERPETRATORS AND THE IMPORTANCE OF HUMAI RIGHTS	O .N
Why France, Germany, and Belgium are chosen for comparison	22
The explanatory reasons explaining why German parliaments approved the arms exports to the third countries identified as human rights violators.	he 23

The explanatory reasons explaining why French parliaments approved the arms exports to the third countries considered as human rights perpetrators	
The explanatory reasons explaining why Belgian parliaments approved the arms exports to th	
third countries considered as human rights perpetrators	25
The explanatory reasons explaining why Dutch parliaments approved the arms exports to thir countries considered as human rights perpetrators	
THE ROLE OF INTEREST GROUPS IN FRANCE, GERMANY, BELGIUM, AND THE NETHERLANDS IN INFLUENCING THE PARLIAMENT'S APPROVAL	28
CHAPTER 3 RESEARCH METHOD AND SOURCES	29
CHAPTER 4 RESEARCH RESULTS	34
CHAPTER 5 CONCLUSION	39
REFERENCE	40

Abbreviations

ECC: European Code of Conduct

EC: European Commission

EU: European Union

WA: Wassenaar Arrangement

Number of words (content): 20.227

Pandu Wijaya (s1203991)

THE NATIONAL PARLIAMENTS' APPROVAL ON THE EU MEMBER STATES' ARMS EXPORTS

Abstract

This thesis explains the arms export control in EU Member States after the establishment of the European Code of Conduct (ECC)'s criteria, where the need to protect human rights is incorporated as one of its criteria. This thesis shows that, although EU Member States should only control their arms exports using the ECC's criteria, the result of the controls on the exports of arms to third countries considered as human rights perpetrators in different Member States is not harmonized. This is because national parliament of each EU Member State might approve the exports of arms in their countries based on different reasons. Parliaments in the Member States that have more respect to human rights, like the Netherlands, can prevent their governments from granting the permits to the exports that carry risk of violation of human rights. The parliaments in the Member States that have less respect on human rights can still allow their governments to grant the permit for such exports because for them export revenues and bilateral relationship might be more important than human rights. This thesis shows that the national parliaments can control the exports of arms in their countries and prevent their governments from granting the permits to the exports that do not satisfy their wishes.

Key words: parliaments, governments, approval, public value advocating the need to protect human rights, human rights, arms exports, permits, and ECC

•

CHAPTER 1 INTRODUCTION

THESIS AIM

This thesis explains that the arms export control on the exports of arms to third countries considered as human rights perpetrators in different EU Member States is not harmonized. National parliament of each EU Member State might approve the exports of arms to such countries based on different reasons. The purpose of this thesis is to observe the explanatory reasons explaining why national parliament of a Member State approves the attempt of their government to grant the permit to an arms export to a third country considered as human rights perpetrator.

For this purpose, this thesis chooses to observe the explanatory reasons explaining why Dutch parliaments approve the governments' attempts to grant the permit to an arms export to a third country considered as human rights perpetrator. The choice is motivated by the fact that Dutch parliament has ever disapproved the attempt of its national government to grant the permit to an export of tanks to Indonesia in 2012. This parliament's reaction was motivated by the consideration that Indonesia has been identified as human rights perpetrator. Because of this parliamentary objection, the permit for the export of these tanks was not granted. This fact shows that national parliaments in the Member States that have respect to human rights, like the Netherlands, can prevent their governments from granting the permits to the exports that carry risk of violation of human rights.

Another argument that makes the parliamentary control in Netherlands is interesting to study, is the fact that Dutch parliaments allowed governments to grant the permit to the exports of corvettes to Indonesia, but not to the export of tanks. Choosing the country of Indonesia as the recipient country and using these exports phenomena for observation enable this thesis to find out the explanatory reasons behind the parliaments' approval on arms exports. These explanatory reasons help this thesis explain why sometimes parliaments give their approval, and why sometimes not although the recipient country is the same.

To show that the parliaments in other Member States may grant their approval based on other explanatory reasons, this thesis provides literature review describing the explanatory reasons explaining why parliaments in other Member States, like France, Germany, and Belgium approve the attempts of their governments to grant the permits to the exports of arms to third countries considered as human rights perpetrators like Saudi Arabia. This literature review also explains that these countries can be argued to have less respect on human rights compared to the Netherlands. This thesis shows that national parliaments from EU Member States that have less respect on human rights approve the attempts of their governments to grant the permits based on other explanatory reasons.

At the same time, this thesis shows that parliaments in the EU Member States that respect human rights like the Netherlands will only approve the attempts of their governments to grant the permits to the exports of arms to third countries considered as human rights perpetrators like Indonesia when they understand that the exported arms would not be used to violate human rights in recipient countries. The fact that the exported tanks would have more likeliness to be used to violate human rights in Indonesia compared to the exported corvettes, assumed based on their technical use, explains why Dutch parliaments approved the export of corvettes but not the export of tanks. This all

and all shows that the explanatory reasons explaining the parliament's approval on the government's attempt to grant the permit to an arms export to a third country considered as human rights perpetrator are influenced by their respect on human rights. This thesis shows that, because the respect on human rights across the EU is not the same, the arms export control in different Member States is not harmonized, even after the establishment of the ECC's criteria.

How governments' attempts earn parliamentary approval

The citizens provide approval on a particular governments' attempt by submitting their wishes related to that attempt through their interest groups. These interest groups will then forward these wishes to parliamentary members (the political actors representing political parties). These wishes describe the requirements behind their willingness to support an arms export. The parliamentary members use these submitted wishes to ensure that the governments' attempts to grant the permits are not in contrary with these wishes (see Bromely, 2008, p: 34).

The decision of parliaments to disapprove or to approve a particular arms export is influenced by how hard these interest groups lobby them. The lobby of the interest groups is one of the most important factors that move the parliaments (Gilens and Page, 2014).

Interest groups in France, Germany, Belgium and the Netherlands work using transnational framework to link the wishes of their citizens with the interest of the EU as a whole and the interest of the citizens of other Member States and the third countries. Such transnational framework is being established from their networks with the interest groups from other Member States and third countries, their national political parties and the national political parties from other EU Member States (see Beyers, J., and Kerremans, B., n.d.).

The networks that interest groups have with political parties are useful to lobby them, including their members sitting in parliaments and in governments. The political parties, parliaments and governments from France Germany, Belgium, and the Netherlands welcome these interest groups to make contacts with them and to build networks with them, even for those having opposite interests. This is because they want to know about the wishes of their citizens, including their contradictory wishes. From this welcoming attitude, this thesis argues that parliaments and governments in these four countries will always perform in accordance with the wishes of their citizens. In other words, the permits that have been granted in each of these four countries have been in accordance with the wishes of the citizens and their parliamentary representatives (Beyers, J., and Kerremans, B., n.d.).

In the Netherlands, parliaments control the exports of arms to third countries considered as human rights perpetrators. When parliaments do not agree with a particular arms export, parliaments will prevent the governments from granting the permit to that particular arms export. The governments usually take into account this parliaments' position before granting the permit (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesie, Beantwoording vragen van het lid El Fassed over berichten dat Indonesie mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesie (33000-X, nr. 47), n.d., "Handelingen")

(Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

RESEARCH QUESTION

To understand why Dutch parliament approved the government's attempt to grant the permit to the export of corvettes to Indonesia, but not the attempts to grant the permits to the exports of tanks to the same country, the main intention of this study will be to answer the following inquiry:

What are the main explanatory reasons explaining the parliaments' decision to approve the governments' attempts to grant the permits to the exports of arms to Indonesia?

THESIS LIMITATIONS AND THE FOCUS OF THE STUDY

This thesis uses this research question to observe the possible reasons explaining why parliaments approved the governments' attempts to grant the permits to the exports of arms to Indonesia. This thesis in any case only considers that this parliaments' approval is based on good consideration and good information. However this thesis still acknowledges the fact that parliaments might have limited of knowledge because of their dependency on the information submitted by governments. This thesis is not focused on discussing whether the governments always submit honest and clear information to parliaments (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020), 2004) (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen Indonesië EKV en Wapenexportcontrole, 2006) (Tweede Kamer der Staten-Generaal, Exportkredietverzekering, sondages en wapenexportvergunningen, 2005).

This thesis is also not focused on discussing whether parliaments in EU Member States, like the Netherlands, have exercised their powers efficiently (see Klopman, 2012). This thesis is not focused on evaluating parliaments' performances. This thesis is not focused on discussing whether the parliaments have deployed different measures to earn support from the citizens for supporting their motions so that governments will respect their motions. For example this thesis is not focused on discussing whether the parliaments have enhanced the transparency of arms export controls and the exported arms, so that they can give more pressure to the governments to relinquish their attempts to grant some permits (see Klopman, 2012, p: 23) (see Steert, 2009, p: 9-13) (Bono, n.d.).

This thesis acknowledges that the power of the parliaments in different Member States to scrutinize the governments' performances is not the same. This could be seen from the fact that Dutch and German parliaments have stronger powers in commenting on the created Green/White Papers and scrutinizing the position of their countries within the EU policies construction compared to French and Belgian parliaments. However, this thesis is not focused on discussing how strong the parliaments' powers in each EU Member State are in ensuring proper implementation of the ECC's criteria in the arms export control in their countries (see Jensen and Martinsen, 2014).

Moreover this thesis is not focused on discussing whether there has been transparency in the information on the governments' attempts to grant the permits for arms exports and whether there have been effective measures deployed to enhance the transparency of this information. Due to the establishment of "European Administrative Space" (EAS), each EU Member State is expected to adopt the principles of openness and transparency in their public administrations. The principle of openness

suggests that the administration should be made become publicly available. The principle of transparency suggests that the administration of each Member State should enable the public to examine it and evaluate it for the purpose of scrutiny and supervision. Due to the requirement to adopt these principles in the public administration, the parliaments and the governments of each Member State must have published all information linked to the arms exports in their countries so that their citizens can access them and can examine and evaluate them so that they can make a decision on whether they want to delegitimize particular arms exports. For the case of arms exports in the Netherlands, is supported by the fact that according to the Article 2 of the "Wet openbaarheid van bestuur" in the Netherlands, every government organ, including the governments and the parliaments, is obliged to publicize the information that the citizens need to know. These government organs have also the obligation to ensure that the publicized information is actual, correct, and comparable. According to the Article 8 from the same regulation, these government organs with their own manners in any case must ensure that the publicized information advocate the public value of democracy. The Article 8 also regulates that these government organs should ensure that the publicized information is understandable so that the citizens can use it and take advantages out of it in accordance with their needs and can understand how these organs perform and operate. From here it is clear that it can be considered that Dutch parliaments and Dutch governments must have published all information linked to the arms exports in the Netherlands, including the governments' attempts to grant the permits for arms exports, that can be used by the Dutch citizens to make decision on whether they want to legitimize or delegitimize some of those attempts. From this consideration, this thesis argues that, when there is no specific parliamentary motion specifically created to avoid a particular arms export, there is no (strong) interest from the citizens to delegitimize any governments' attempts to grant the permits for any arms exports (see Bustraan, 2012) (see Overheid.nl, wet openbaarheid van bestuur, n. d.) (OECD, 1999).

Although this thesis is aware that the salience of the arms export as an issue for debate in France, Germany, Belgium, and the Netherlands influences the eagerness of the citizens in these four countries and their interest groups to submit their wishes to approve or disapprove particular arms exports to the parliaments so that those exports can be avoided, this thesis is also not focused on discussing whether arms export is an interesting issue for the citizens in France, Germany, Belgium, and the Netherlands that stimulate them to submit to their interest groups their interest to approve and to disapprove a particular arms export (see Beyers, J., and Kerremans, B., n.d.).

This thesis is also further aware that such eagerness is also influenced by the distance between the citizens and their interest groups and the distance between these interest groups and political parties and parliaments, this thesis is not focused on discussing to what extent such distance exists in each of these four countries, to what extent each interest group in these four countries have willingness to lobby all of the necessary parties to get their voices heard, to what extent their networks with these parties are efficient and effective, and to what extent their lobbying has been efficient and effective (see Beyers, J., and Kerremans, B., n.d.).

THEORETICAL FRAMEWORK

The study in this thesis is based on a qualitative analysis on research materials. This qualitative analysis is done using the theoretical framework that includes some scholarship theories necessary to enable the analysis producing theoretically explanations and arguments (see Van Evera, 1997).

The theoretical framework is used to help this thesis investigate and analyze qualitatively the causes and effects in the phenomena linked to the parliamentary control on arms exports, including the explanatory reasons explaining why the parliaments approve an arms export (see Van Evera, 1997). Some scholarship theories concerned on the parliaments' approval on the governments' attempts are used in this theoretical framework (see Van Evera, 1997).

These scholarship theories explain that parliament will approve a government's attempt when government respects the existing law, shares the motive behind that attempt with the parliament and respects the public values of their citizens, including the public value protecting human rights (see Pfaff, 2011) (see Schmelzle, 2011).

Using these theories as conceptual framework, this thesis observes qualitatively how the government's respect on the existing law like the ECC's criteria matters. Because Member States are obliged to implement the ECC's criteria during the application assessment, this thesis seeks to investigate whether Dutch parliament disapproved the government's attempt to grant the permit to the export of tanks to Indonesia because government had not implemented the ECC's criteria when assessing the permit application. To find out whether this incompliance was indeed the reason, this thesis also investigates whether governments had fulfilled that obligation for the export of corvettes (see Pfaff, 2011) (see Schmelzle, 2011).

Using these theories as conceptual framework, this thesis observes qualitatively how the government's willingness to share the motive behind their attempt matters. This thesis investigates whether the fact that government had not shared the (economic) motive behind the export of the tanks had been another explanatory reason why parliament disapproved their attempt to grant the permit for the export of those tanks. To find out whether this was indeed the reason, this thesis also investigates whether governments had shared the motives behind the exports of the corvettes (see Pfaff, 2011) (see Schmelzle, 2011).

Using these theories as conceptual framework, this thesis observes qualitatively how the government's respect on the public values of their citizens, including the public value protecting human rights, matters. This thesis investigates whether one of the explanatory reasons why the parliament disapproved the government's attempt to grant the permit to the export of tanks was the fact that the government had forgotten to assess the risk of violation of human rights that the export of tanks carries. To find out whether this was indeed the reason, this thesis also investigates whether Dutch governments had not forgotten to do such assessment for the export of the corvettes (see Pfaff, 2011) (see Schmelzle, 2011).

Furthermore, using these theories as theoretical framework, this thesis also observes qualitatively the explanatory reason that explains why parliament decided to approve the government's attempt to grant the permit to the exports of corvettes but not to the exports of tanks (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (Kumm, 2004) (Schmelzle, 2011) (Pfaff, 2011) (Van Evera, 1997).

RESEARCH METHOD AND SOURCES

The research design in this thesis is based on case study research where one case is deeply observed to provide a picture of phenomena linked to the parliamentary control at the national level in an EU

Member State provided on the exports of arms to third countries considered as human rights perpetrators. The parliamentary control provided in the Netherlands on the exports of arms to Indonesia is chosen as the case to study (see Sagepub, n.d.) (see Thomas, 2003).

Besides observing the parliamentary control provided in the Netherlands on the exports of arms to Indonesia, to provide a an overview on the parliamentary control in other EU Member States, this thesis provides a comparative analysis of NL, Germany, France and Belgium. France, Germany and Belgium are chosen for comparison due to their arms exports to Saudi Arabia, which has also been identified as human rights violator. The fact that they export arms to Saudi Arabia shows that their respect on human rights is less compared to the Netherlands. Another reason why France, Germany, and Belgium are chosen as comparative countries instead of other EU arms exporters like United Kingdom, Sweden and Italy is the consideration that there is a bigger possibility that the Netherlands mimics arms exports decisions of France, Germany, and Belgium rather than the exports decisions of United Kingdom, Sweden, and Italy. This is due to the argument that the Netherlands has shared more historical paths with France, Germany, and Belgium than with United Kingdom, Sweden and Italy. And one reason that countries usually have to mimic foreign policies of other countries is its sharing of historical paths, identities, cultures, values, networks and dependencies grounded in its historical relationship with those countries (see Pauly Junior, 2013) (see Soetendorp, 2014) (see Lak, 2011) ("Faithfull Allies?", n.d.) (Bromley, 2012) (Willardson, 2013) (Dodd, Lyklema, and Van Weringh, 2006) (Grebe, J., 2014) (see Poullie, 2014, p: 39) (Detjen, J., n.d.) (Tacq, 2009) (Haspeslagh, M., 2011) (Duquet, 2014).

The overview on the parliamentary control in France, Germany and Belgium provided on the exports of arms to the third countries identified as human rights violators like Saudi Arabia is provided from some scholarship literatures available online. These literatures provide brief information explaining why the parliaments in these Member States approve the attempts of their governments to grant the permit to the exports of arms to third countries considered as human rights perpetrators. These literatures also provide brief argument that one explanatory reason why these parliaments approve such exports is the consideration that the respect on human rights in these countries is not that high (see Sagepub, n.d.) (see Thomas, 2003).

To study the parliamentary control provided in the Netherlands on the exports of arms to Indonesia, this thesis analyze qualitatively the content of the parliaments' and governments' letters submitted since 1998 until 2013. The year of 1998 is chosen because it was the starting point of the obligation to adopt the ECC's criteria. The year of 2013 is chosen due to the need to find the needed and accessible research materials. These parliamentary letters are the letters that parliaments have sent to the Ministries asking some questions on some arms export issues. The letters that governments submit to parliaments are the letters replying these letters providing information that has been asked by the parliaments. Basically these governments' letters explain that they have implemented the ECC's criteria during the application assessment, that there have been good reasons to grant permits for some submitted applications and that the need to protect human rights has been taken into account during the application assessment (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (see Sagepub, n.d.) (see Thomas, 2003).

By providing such explanation to parliaments, through their letters governments want to motivate parliaments to approve their attempts to grant the permits. On the other hand, through the

parliaments' letters parliaments inform whether they can accept and agree with the governments' arguments (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (see Sagepub, n.d.) (see Thomas, 2003).

THESIS OUTLINE

This thesis is divided into five chapters. The first chapter, which is this chapter, provides information about the aim of this thesis, the research question of this thesis, short literature review, conceptual framework and a short description on the research method and sources used in this thesis. The literature reviews in this chapter explains shortly how the governments' attempts granting the permit to arms exports earn parliamentary approval.

The second chapter provides different literature reviews. These literature reviews include the literature review on the ECC, which is the control instrument that each Member State should implement when assessing the permit applications. This review provides a short description about the impact of the obligation to implement the ECC's criteria when controlling arms export in the Netherlands.

Another literature review in Chapter two is the literature review on the history of the regulations for the arms export control regulations in the Netherlands. This literature review is there to provide an insight about how is the history of the development of the arms export control in the Netherlands and the development of the regulations regulating such control. This literature review explains the development of the arms export control in the Netherlands before and after the introduction of the ECC's criteria.

The next literature review is about how the control provided on the exports of arms happens in practice in the Netherlands. This literature review provides brief explanation about how the arms exports to third countries are being controlled in the Netherlands.

The next literature review from chapter two is the granting mechanism of the arms export permit. This literature review provides information about the granting permit mechanism, the actors involved, and their administration responsibilities linked to export and import field.

Chapter two also explains the role of the Dutch parliaments in the granting of the permit. Chapter two argues that, although the parliaments do not have the authority to make the decision to grant the permit or the denial permit, their approval on the governments' attempts to make those decisions is very much taken into account by the Dutch governments when assessing the permits applications.

Following this argument Chapter two also explains why the governments consider that such parliamentary approval is important. Chapter two explains that this approval is important because the parliaments are there to represent the citizens. The governments need to have their approval to ensure that their decision to grant the permit for a particular export will be accepted by their citizens. The citizens' acceptance on the governments' decision is important to increase the effectiveness of their governance.

Chapter two is also there to provide short literature reviews on the scholarship theories explaining some explanatory reasons why the parliaments approve the governments' attempts to make a

particular decision, including their attempt to grant the permit for a particular export. This thesis uses these theories in its theoretical framework to investigate whether these reasons might have been the reasons explaining why the Dutch parliaments only approved the governments' attempts to grant the permits for the exports of corvettes to Indonesia, but not the tanks.

Chapter two also provides some literature reviews on the explanatory reasons explaining why the attempts of the governments to grant the permit can earn approval from the parliaments in other Member States. This chapter provides some reviews on the explanatory reasons explaining why the parliaments in some Member States like France, Germany and Belgium allowed their governments to grant the permits for arms exports to the third countries identified as human rights violators. These reviews provide an insight about whether implementing the ECC's criteria when assessing the permit applications is necessary for earning the parliamentary approval. These reviews also explain how important the need to respect human rights is for earning the parliamentary approval. Furthermore, these reviews provide an insight that for some EU Member States like France, Germany, and Belgium these two needs are less important compared to other needs, such as an economic incentive from the exports and an improvement of bilateral relationship. In this section, it is also explained why the countries of France, Germany, and Belgium are chosen as comparative countries instead of other big arms exports like United Kingdom, Sweden, and Italy.

In regards to these reviews, before researching about the explanatory reasons explaining why the Dutch parliaments approved the arms exports to Indonesia, in Chapter two this thesis provides some literature reviews on how important the public value respecting human rights is in the Netherlands. This importance explains why the parliaments disallowed the governments to export tanks to Indonesia.

At the end, the literature reviews also provide some reviews on the interest groups in France, Germany, Belgium and the Netherlands, especially on the influence of their networks on the restrictiveness of national parliament on controlling their country's arms exports.

In chapter three this thesis explains about the research method which is used in this thesis to investigate about the explanatory reasons behind the parliaments' approval on the governments' attempts granting the permits for the exports of arms to Indonesia. Chapter three also explains the research sources used in this thesis.

Chapter four provides the research results of this thesis. These results describe that the explanatory reasons explaining why parliaments approved the exports of corvettes to Indonesia but not the export of tanks. These research results also explain that the seriousness of the governments' respect on the public value protecting human rights is the most important source to earn approval from parliaments on their attempts to grant the permits.

Chapter 5 provides conclusion of the research results and a brief suggestion for the next research.

CHAPTER 2 LITERATURE REVIEW

THE MOTIVES BEHIND THE ESTABLISHMENT OF THE ECC

The exchange of the real-time information of the permit denials among control officials in different Member States is the most important reason that makes the ECC offer a better control mechanism compared to the Wassenaar Arrangement (WA). In addition, the ECC also includes less number of countries engaged compared to the WA. This less in number of the countries involved reduces the diversity of wishes (Bromley, 2008, p: 38).

There are three driving factors that motivate the harmonization of the arms export controls in different Member States. The first driving factor is the consolidation and internationalization of the arms industries. Some arms industries in different Member States decide to consolidate with each other so that they can export their arms from different Member States. Because of this, the EU Member States agree with each other that they should apply similar controls on the exports of these industries (see Smith, K., E., 2001 p: 187 cited in Bromley, 2008, p: 5).

The second driving factor is the need to harmonize the foreign and security policies from different Member States. This is due to their interest to avoid war among them (see Smith, K., E., 2001 p: 187 cited in Bromley, 2008, p: 5).

The third driving factor is the need of different Member States to harmonize their attempts to create a common position towards third countries. Such a common position is necessary for example when they aim to apply embargo together towards a particular third country (see Bromley, 2008, p: 7-8).

THE EFFECTIVENESS OF THE ECC

The responsibility to make the decision to grant the permit still belongs on the hands of each Member State. The European Commission (EC) has no substantial power to enforce similar implementation of the ECC's criteria during the control in different Member States. Different Member States can grant different permit decisions on the similar arms exports (see Bromley, 2008, p: 10-13) (see Bromley, 2008, p: 14-16).

THE IMPACTS OF THE OBLIGATION TO IMPLEMENT THE ECC'S CRITERIA IN THE ARMS EXPORT CONTROL MECHANISM IN THE NETHERLANDS

The obligation to implement the ECC's criteria when assessing the permit applications does not change the decision making procedures in the government level and in the parliamentary level that much (Bromely, 2008, p: 37). There is also no impact on the day-to-day control activities (Bromely, 2008, p: 37) However, the exporters argue that the Netherlands is more restrictive in granting the permit compared to other Member States (Bromely, 2008, p: 33) (Bromely, 2008, p: 37).

THE HISTORY OF THE DUTCH REGULATIONS FOR THE ARMS EXPORT CONTROLS IN THE NETHERLANDS

The "Algemene wet" established in 1822 was the first Dutch law regulating the export controls in the Netherlands. This law regulated the controls for the import, export and transit of products in the Netherlands. In this law, the transactions for arms were treated like the transactions for other products. There was no special law for the control of arms transfers (Customs, 2008).

The "Algemene wet" was replaced by the "Algemene Wet inzake de Douane en Accijnzen (AWDA)" on 5 July 1962. The AWDA still did not distinguish the transfers of arms and the transfers of other products (Ministry of Economic Affairs, 1962 cited in Lagerwaard, 2009, p: 26).

In the same year, in 1962, there was another law created called the "In- en Uitvoer Wet". This law also did not contain specific regulations for arms transfers (Customs, 2008 cited in Lagerwaard, 2009, p: 26).

It was just on 26 April 1963 a law containing specific regulations regulating arms transfers for the first time was created. This law is called the "In- en Uitvoer Besluit Strategische Goederen". According to this law, arms exporters must acquire the permit from the Ministry of Economic Affairs before transferring their arms (Ministry of Economic Affairs, 1963 cited in Lagerwaard, 2009, p: 26).

To complete the "In – en Uitvoer Besluit Strategische Goederen", another law called the "Sanctiewet" was established in 1977. The motif behind its establishment was to accommodate the interest to harmonize the national decisions of different countries on applying an (international) economic sanction against a particular country, e.g. the decision to create an embargo against particular third countries (Ministry of Foreign Affairs, 1977 cited in Lagerwaard, 2009, p: 26).

The "Wet Wapens en Munitie (Weapons and Ammunition Act)" came into force on 16 November 1995 and was established to prohibit, and therefore regulate through permits, activities dealing with weapons and ammunition, including manufacturing trading, importing, exporting, possessing, carrying and etcetera. In terms of exporting and importing arms, the permit applications were applied to the Ministry of Economic Affairs and the granted permits were checked when the arms were passing the Customs. The law was being replaced by another law on 1 January 1997 called the "Herziening Wet Wapens en Munitie" (Ministry of Justice, 2003 cited in Lagerwaard, 2009, p: 27).

The "Algemene Douanewet (General Customs Act)" came into force on 1 August 2008 replacing the "In- en Uitvoer Wet" and the "Douanewet". The "Algemene Douanewet (General Customs Act)" adopts the European law, including the ECC and its criteria (Customs, 2009a, Lagerwaard, 2009, p: 27).

This part explains that the first Dutch law providing special regulations to regulate arms transfers was established in the 1960s. Before this period, the arms transfers were treated as the international trades of the common products. The concern of different Member States to keep improving the harmonization of their arms export controls has been one of the reasons motivating the Dutch governments to keep developing their control regulations and improving their controls on permit applications (see Lagerwaard, 2009).

THE ARMS EXPORT CONTROLS IN PRACTICE IN THE NETHERLANDS

The Dutch Ministry of Economic Affairs has been authorized to assess each permit application and to grant the permit. The Dutch Customs authority controls the arms exports at four Dutch borders, including Rotterdam, North, West and South. At these borders, the Customs authority implements these core duties:

- 1. Stopping the arms passing the borders;
- 2. Ensuring the CDIU controlling the invoice and the granted permit properly. When no permit, the arms cannot be exported. The CDIU should match the physical of the exported arms with the data in their export documents;
- 3. Levying and collecting taxes on the exported arms (see Customs, 2009b cited in Lagerwaard, 2009, p: 27-28).

THE ROLE OF THE DUTCH MINISTRIES IN THE ARMS EXPORT CONTROLS IN THE NETHERLANDS

In this part this thesis explains about the granting permit mechanism, actors are in charged, and their administration responsibilities.

The arms exporters must first submit a permit application to the Centrale Dienst voor In-en Uitvoer (CDIU). Without this permit, these exporters are not allowed to export arms. The CDIU performs on behalf of the Ministry of Foreign Affairs receiving and examining the permit applications. The CDIU assess each permit application to investigate whether each arm receiver country can satisfy all of the ECC criteria or not. The CDIU is in charged to examine the application when the exports are purposed for other NATO Member States, other EU Member States, Switzerland, Japan, Australia, and New Zealand. The CDIU examines those applications using the instructions given by the Ministry of Economic Affairs. When the export is purposed for other countries, the CDIU needs to consult its assessment with the Ministry of Foreign Affairs and The Ministry of International Trade and Cooperation with the Development Countries. The minister of Foreign Affairs provides advice to the minister of Economic Affairs on the security risks and the bilateral benefit that a particular arms export possesses. In examining the information about the technic specifications of the arms proposed for export, the CDIU and the Ministry of Foreign Affairs examine whether the information mentioned in the application about the use purpose of the exported arms proposed matches with their technical functions. The recipient countries are the most important for the assessment. One of the information sources is the Dutch embassy situated in the recipient country. The information about the recipient country includes the information on what it does with the received arms, the information on to what extent it has been actively trying to purchase the arms from other countries, and the information on whether another Member States has granted a denial permit on the exported arms. The Ministry of Foreign Affairs also takes into account the information on the nature of the transaction when assessing the permit applications. This information includes the information on the payment and on the shipment. In addition, this information also includes the information on the recipient country, the transit country and the brokers involved. The minister of Foreign Affairs implements the ECC's criteria when assessing the permit applications. The Minister for International Trade and Cooperation with the Developing Countries helps the Ministry of Foreign Affairs provide advices on the bilateral risks that permit denials might produce. When the exports are purposed for third countries, besides the Ministry of Economic Affairs, these Ministries are responsible for the granting of the permit. The CDIU is still in charged in handing over the permit after the approval from these Ministries is granted (Rijksoverheid.nl, "Regels voor export en doorvoer van militaire goederen", n. d.) (Ministerie van Buitendlanse Zaken, 2013.122971, 2013, p: 1-2) (Tweede Kaer der Staten-Generaal, Wapeneportbeleid 22 054, brief nr. 217, 2013) (In- en Uitvoerwet 1944 cited in Overheid.nl, Wet- en regelgeving, Beleidsterrein In- en uitvoerregelingen, 2014) (Douane Belastingdienst, 110.00.18, n.d.) (Ministerie van Buitendlanse Zaken, 2013.122971, 2013, p: 19) (Overheid.nl, Wet- en regelgeving, Beleidsterrein In- en uitvoerregelingen, 2014). (Ministerie van Buitendlanse Zaken, 2013.122971, 2013, p. 3) (Ministerie van Buitendlanse Zaken, 2013.122971, 2013, p. 3) (Ministerie van Buitenlandse Zaken, 2013, p: 33) (Overheid.nl, Wet- en regelgeving, Beleidsterrein In- en uitvoerregelingen, 2014) (Uitvoerbesluit Strategische goederen 1963 cited in Overheid.nl, 2014).

When the assessment produces a desired result, the permit will be granted, otherwise not. When the application is rejected, the importer and recipient country receive a refusal notification including the

reason behind the rejection. Such rejection will be shared immediately to other EU Member States. Other EU Member States attempting to grant the permits on the rejected applications must consult their attempt to those rejecting Member States before granting the permits (Ministerie van Buitendlanse Zaken, 2013.122971, 2013, p: 3).

THE ROLE OF DUTCH PARLIAMENTS IN THE ARMS EXPORT CONTROLS IN THE NETHERLANDS

Dutch parliaments apply control on the arms exports in the Netherlands and grant their approval based on the information from different sources, including from governments. This information includes the information on the export credit insurance, the information on the arms production, the information on how the arms producers let the country pay the loss when the permits are not granted (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020), 2004) (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen Indonesië EKV en Wapenexportcontrole, 2006) (Tweede Kamer der Staten-Generaal, Exportkredietverzekering, sondages en wapenexportvergunningen, 2005).

This information is usually provided through the ministerial letters submitted to parliaments. In these letters the Ministers usually provide this information using causal analysis to provide better understanding about the background of each issue. For example, in explaining about the need to use export credit insurance, governments explained that, because there is no certainty that a particular export permit application will not be refused, arms producers use the export credit insurance to cover the possible loss, produce their products based on orders and submit their permit applications after the production is finished. Furthermore, governments also explained that, such export credit insurance is not taken directly by the producer, but by the banks financing its arms production instead. This implicitly means that, because this insurance is financed by the country, when the permit is not granted, the country pays the insurance loss covering the production expenditures (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020), 2004) (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen Indonesië EKV en Wapenexportcontrole, 2006) (Tweede Kamer der Staten-Generaal, Exportkredietverzekering, sondages en wapenexportvergunningen, 2005).

For the exports to third countries considered as human rights perpetrators, Dutch parliaments apply a more stringent control. Assessing the permit applications based on the ECC's criteria, including its criterion protecting human rights, may not always be enough to show to parliaments how serious governments have taken into account the need to avoid any risk of violation of human rights that an export carries. This means that informing parliaments that they have assessed the permit application using the ECC's criterion protecting human rights does not always enable them to earn approval from parliaments on their attempt to grant the permit for the application assessed (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

WHY PARLIAMENTS' APPROVAL IS IMPORTANT

Most governments take into account the opinion of their parliaments. Dutch governments usually take into account the parliaments' decision to disapprove particular arms exports. They do so because the parliamentary approval gives a long term impact on the effectiveness of their governance. This is based on the argument that the parliaments' approval on the governments' attempts to make a particular decision, like granting the permit, is important to enable the governments to earn citizens' acceptance on that particular decision after it is made. This citizens' acceptance is necessary to increase their respect on the decision taken. Such respect is important for the effectiveness of the governments' governance. Governance effectiveness is argued to be consequence of citizens' acceptance on governments' decisions (see Weber 1978, p: 212ff cited in Schmelzle, 2011, p: 8) (see Levi/Sacks 2009 cited in Schmelzle, 2011, p: 12) (Schmelzle, 2011, p: 13) (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020), 2004) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

THE SOURCES TO EARN PARLIAMENTS' APPROVAL

There are some explanatory reasons explaining why parliaments approve the governments' attempts to grant the arms export permits. Some of these reasons include the government's respect on the existing law, the government's willingness to share the motives behind their attempts, and the government's respect on the citizens' public values, including the public value protecting human rights.

According to legal scholars, parliaments will approve the governments' attempts to grant the permit, when those attempts are in accordance with the ECC's criteria. Legal scholars argue that one important factor to earn parliamentary approval is the legality background. This thesis observes whether the government's incompliance with their obligation to implement the criteria of the ECC's criteria when assessing the permit application for the export of tanks to Indonesia in 2012 was one of the explanatory reasons explaining why parliament disapproved the government's attempt to grant the permit for the export of those tanks. To find out whether this incompliance was indeed the reason, this thesis also investigates whether governments had fulfilled that obligation for the exports of the corvettes (Pfaff, 2011, p: 4).

Furthermore, some scholars also argue that parliaments approve the government's attempts to grant the permits when governments share the motives behind their attempts. This motive sharing provides an opportunity for parliaments to understand more about the goals that those attempts aim to achieve and the impacts of their approval on the achievement of those goals. In regards to this point of view, this thesis observes whether the fact that government had not shared the (economic) motive behind the export of the tanks had been another explanatory reason explaining why parliament disapproved the government's attempt to grant the permit for the export of those tanks. To find out whether this was indeed the reason, this thesis also investigates whether governments had shared the motive behind the exports of corvettes (see Schmelzle, 2011, p: 14).

According to scholars, using the content-dependent concept, parliaments approve the government's attempts to grant the permits when these attempts respect the citizens' public values. These public values include the public value protecting human rights which is very important for the Dutch citizens. Parliaments approve the government's attempts when they consider that the "content" of these attempts does not violate the citizens' public values. This thesis observes whether one of the explanatory reasons explaining why parliament disapproved the government's attempt to grant the permit for the export of tanks to Indonesia was the fact that the government had forgotten to assess the risk of human rights violation that this export carries. One way to assess such a risk is by assessing the application for this export using the ECC's criterion protecting human rights. This thesis investigates whether parliaments decided to disapprove the governments' attempt to grant the permit for this export because government had forgotten to implement the ECC's criterion protecting human rights when assessing the application for this export. To find out whether this was indeed the reason, this thesis also investigates whether governments had not forgotten to do such assessment for the export of corvettes (see Schmelzle, 2011, p: 10-11).

The incorporation of the public value protecting human rights in the preamble and Article 1 of the Universal Declaration of Human Rights (UDHR), and the establishment of the ECC's criterion advocating the need to protect human rights produce an obligation for every UN Member States and the EU Member States, to respect and protect human rights (General Assembly, 1948 cited in Ericsson, 2008, p: 8) (Ericsson, 2008, p: 8-9) (Hansen and March, 2013). In regards to this obligation, the following section provides literature review on the explanatory reasons explaining why parliaments in other Member States like France, Germany and Belgium approved the attempts of their governments to grant the permit for the arms exports to third countries considered as human rights perpetrators, including Saudi Arabia (see Ericsson, 2008).

EXPLANATORY REASONS EXPLAINING WHY PARLIAMENTS IN OTHER MEMBER STATES APPROVE THE ATTEMPTS OF THEIR GOVERNMENTS TO GRANT THE PERMITS TO ARMS EXPORTS TO THIRD COUNTRIES CONSIDERED AS HUMAN RIGHTS PERPETRATORS AND THE IMPORTANCE OF HUMAN RIGHTS

To provide a better picture on the parliamentary control across the EU provided by national parliaments on the exports of arms to third countries considered as human rights perpetrators, this section provides review on the explanatory reasons explaining why parliaments in EU Member States, like France, Germany, and Belgium approve the attempts of their governments to grant the permits to the exports of arms to such countries.

Because every Member State is required to respect human rights, in this section, this thesis provides literature review explaining whether national parliaments in different Member States like France, Germany, and Belgium grant their approval on the governments' attempt to grant the permit based on the concern to prevent the risk of violation of human rights that an export carries. The review explains that in France, Germany, and Belgium, national parliaments approve the governments' attempts to grant the permits to the exports of arms to third countries that have some human rights issues like Saudi Arabia. The review shows that the public value protecting human rights for the parliaments in these Member States seems to be somewhat not that important and this is because

this public value is not rooted in their societies. This rooting explains how important this value is for their citizens. When this public value is important for their citizens, parliaments will put more concern to increase the governments' respect on this value. The literature review also explains that the importance of this value for the citizens can be traced from the legitimate laws and constitution. The literature review includes the review on some laws and constitutions of these EU Member States. This review is used to argue to what extent this value is rooted in their societies. Furthermore, the review shows that national parliaments in such Member States do not grant their approval based the need to respect human rights, but based on other reasons. This thesis argues that parliaments from the Member States that have less respect on need to protect human rights put less concern on the risk of violation of human rights that an export carries, and therefore they approve the exports based on other reasons (Duquet, 2014) (see Poullie, 2014) (Grebe, J., 2014) (Tacq, R., 2009, p: 28-32) (Detjen, J., n.d.) (Haspeslagh, M., 2011) (Duquet, 2014, p. 40-52) (see Schmelzle, 2011) (see Pfaff, 2011) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (General Assembly, 1948 cited in Ericsson, 2008, p: 8) (Ericsson, 2008, p: 8-9) (Hansen and March, 2013).

Why France, Germany, and Belgium are chosen for comparison

Besides due to their arms exports to third countries considered as human rights perpetrators, these countries are also chosen for comparison due to some other arguments. The first argument is the argument that countries sometimes mimic the foreign policies of other countries, including the decisions linked to those policies. And because the arms export is a tool part of foreign policy of a country, the decision to export arms that a country has is linked to its foreign policy. Therefore when a country mimics foreign policy of another country, it can be argued that it will make its arms export decision in accordance with the arms export decision of the country it mimics. These countries are chosen as comparison instead of other Member States because of the possibility that the Netherlands might mimic the foreign policies of these countries is bigger compared to other Member States like United Kingdom, Sweden, and Italy. This is because the Netherlands has shared more historical paths with France, Germany, and Belgium than with United Kingdom, Sweden and Italy. And one reason that countries usually have to mimic foreign policies of other countries is its sharing of historical paths with these countries. This historical paths include; identities, cultures, values, networks and dependencies, all and all grounded in its historical relationship with these countries (see Pauly Junior, 2013) (see Soetendorp, 2014) (see Lak, 2011) ("Faithfull Allies?", n.d.) (Bromley, 2012) (Willardson, 2013) (Dodd, Lyklema, and Van Weringh, 2006) (Detjen, J., n.d.) (Poullie, 2014) (Grebe, 2014).

The explanatory reasons explaining why German parliaments approved the arms exports to the third countries identified as human rights violators

The amount of exported arms from Germany to the third countries that have been identified as human rights violators has been increased. Such third countries include Algeria, Indonesia, Qatar and

Saudi Arabia. This might one wonder why German parliaments approved the governments' attempt to grant the permits for the exports to such countries (Poullie, 2014, p. 3).

One example of those exports is the export of patrol boats to Saudi Arabia secured by export credit insurance totaling Euro 1.4 billion. Other deals include the deal for battle tank with Qatar, the deal for military equipment with Algeria and the deal for tanks with Indonesia. The tanks deal with Indonesia might be one of the most controverstial examples showing that German parliaments make different judgement on the need to prevent violation of human rights in recipient countries compared to Dutch parliaments (Grebe, J., 2014) (see Poullie, 2014, p: 39) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

To understand the explanatory reasons explaining why German parliaments approved arms exports to these third countries, one should understand first the German control measures on arms exports. In Germany, arms exports are controlled to ensure whether they meet the German "objectives, interests, and values". This is because Germany links its arms exports to its foreign and security policies (Grebe, 2014) (see Poullie, 2014, p: 39).

The German foreign and security policies, including the decisions on arms exports, are produced based on the German "values, interests and objectives" (Poullie, 2014, p. 39). The German "objectives" include providing security and protection of German citizens, providing territorial integrity and sovereignty for Germany and its allies and fulfilling the international responsibilities, including its responsibilities to deliver liberty and peace. The German "security interests" include preventing and managing conflicts and crises in third countries that may produce security threats for Germany and its allies and advocating and implementing its foreign and security policies using an assertive and credible manner. The German "public values" include the public value of freedom, the public value of safety and the public value of democracy (Detjen, J., n.d.) (Poullie, 2014) (Grebe, 2014).

The public value protecting human rights is just the public value that Germany is interested to promote. This public value is not incorporated in the German interests and objectives. This public value also has not been originally incorporated in the German Constitution. The public values in the Constitution that are linked to this public value were just adopted in the Constitution after the Second World War. The important reason behind their adoption is linked to the need to avoid the human rights violations during the Hitler regime being repeated (Detjen, J., n.d.) (Poullie, 2014) (Grebe, 2014). The fact that this value was not incorporated in the German Constitution since the beginning of its establishment explains that this value is not rooted in the Dutch society. This all and all shows how important the public value protecting human rights for the German society compared to the need to satisfy German "objectives, interests, and values". German parliaments do not grant their approval on the governments' attempt to grant the permit based on whether their arms exports may impose any risk of violation of human rights violation, but on whether they meet the German "objectives, interests and values" (Detjen, J., n.d.) (Poullie, 2014) (Grebe, 2014) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek

aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

The explanatory reasons explaining why French parliaments approved the arms exports to the third countries considered as human rights perpetrators

Similar to Germany, France also considers its arms exports as part of its foreign policy and security policies. This has happened since the president Charles de Gaulle started developing bilateral relationships with other countries to gain influence on them. Since then France has used its arms exports to finance its arms industries to be able to sell more and to gain more influence. For French parliaments exporting arms is part of these policies and has nothing to do with the risk of violation of human rights in recipient countries (Duquet, 2014) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

The unawareness of French parliaments on the need to protect human rights might be explained by the consideration that the public value protecting human rights is not rooted in the French society. This can be seen from French national laws and decisions linked to the need to protect human rights. One example from these laws and decisions is the national law created against burga. France has created a national law that forbids the use of burga in public spaces. In addition to this law, although the freedom of religion has been advocated as a human right that must be respected according to the Universal Declaration of Human Rights (UDHR); the International Covenant on Civil and Political Rights (ICCPR); the European Convention on Human Rights and Fundamental Freedoms (ECHR) and the Charter of Fundamental rights of the European Union, the need to avoid the use of burqa has already been attempted by the French society since 20 years ago. Another example of these national laws and decisions is the national decision to block the adoption of the revised CoC as a Common Position until the end of 2008. The revise of the ECC could not be completed due to the France's objection. Due to the Common Position, when there is an objection of a Member State on a particular decision, the intentions of other Member States to take that decision are blocked, and therefore the attempted decision cannot be taken by the EU. This France's objection is related to the France's request to lift up the arms embargo on China which has also been blocked by other Member States. France has been interested to lift up the international arms embargo on China although France has understood that that embargo is purposed to prevent further human rights violations that the Chinese governments have done. These national law and decision explains that the public value protecting human rights is not rooted in the French society. This all and all explains that, for France, the explanatory reasons why French parliaments approved the French arms exports to the third countries considered as human rights perpetrators only include the need to improve bilateral relationship with those countries and the need of arms exports revenues to sustain its arms industries, and do not include the concern to prevent violation of human rights in recipient countries (Tacq, 2009, p: 28-32) (Drgoncova, J., 2011) (Duquet, 2014, p: 30-40).

The explanatory reasons explaining why Belgian parliaments approved the arms exports to the third countries considered as human rights perpetrators

Belgian national governments and national parliaments have agreed to regionalize arms export control in 2003. Since then the capital of each region including the Flemish, Walloon and Brussels Capital Region have been responsible for controlling arms exports in their regions, and have been authorized in determining, implementing and monitoring the handling of the permits in their regions (Duquet, 2014).

The Walloon and Flemish defense industries produce different products and have different customers. The Walloon defense industry produces products such as firearms, ammunition and explosives. Its customers are end-users across the world. Flemish defense industries produce the components for the production process like radar and communication equipment, visualization screens, vehicle components and imaging equipment. Their customers are industries in EU and North America (Duquet, 2014, p: 40-41).

In the Federal level, the interest of different regions in Belgium to sell arms to Saudi Arabia is motivated by the need to increase their competence to compete with other regions. The regionalization of export control motivates each region to export more arms and decrease their stringent in controlling the exports. The permits seem to always be granted in each region. The Federal (national) level only controls the exports of surplus arms from Defense department. However, such control only involves the representatives of the political parties sitting in the Federal governments, but not in the Federal parliaments. The arms exports to Saudi Arabia have never been discussed in the Federal parliaments. Federal parliaments do not seem to grant their approval on the governments' attempt to grant the permit based on the concern to ensure that the exported arms will not impose any risk of violation of human rights in recipient countries (Duquet, 2014, p. 43-48) (Duquet, 2014) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

Flemish Parliaments seem also to be not that eager to stop the exports of arms to third countries considered as human rights perpetrators like Saudi Arabia. When they knew that Belgian arms might have been re-exported by Saudi Arabia to Syria, they only questioned Flemish regional government whether those arms had been made by Flemish industries. There has not been a parliamentary motion adopted to add Saudi Arabia in the list of "on hold" countries or to propose an embargo on Saudi Arabia. Flemish regional parliaments will only take a more stringent action to control the exports of arms to Saudi Arabia if other EU Member States, other than the Netherlands, do the same. Flemish regional parliaments do not seem to grant their approval on the governments' attempt to grant the permit based on the concern to ensure that the exported arms will not impose any risk of violation of human rights in recipient countries (Duquet, 2014, p: 48) (Duquet, 2014) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de

verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

Walloon region has always been interested to export arms to Arab states despite its human rights issues. The Arab Spring did not reduce this interest. Thus Walloon parliaments also do not seem to grant their approval on the governments' attempt to grant the permit based on the concern to ensure that the exported arms will not impose any risk of violation of human rights in recipient countries (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië). (Duquet, 2014, p: 46-51). Belgian regional parliaments grant their approval based on the need of arms export revenues and the need to compete with other regions, but not based on the concern to prevent violation of human rights in recipient countries (Duquet, 2014, p: 40-51).

To understand how critical Belgian national and regional parliaments might concern about the issues of violation of human rights, one might need to look on the fact how Belgian parliaments have concerned on the need to respect human rights in Belgium itself. After French, Belgium is another country that bans the use of burqa. Belgian government in 2011 introduced a general ban on the wearing of burqa in public places. This law was created against burqa. Before this national law was introduced, in some areas there had already been some local regulations avoiding the use of burqa. From here one can argue that the public value protecting human right (of freedom of religion) might be not rooted the Belgian society. This might be the explanatory reason why Belgian national and regional parliaments approved their governments' attempts to grant the permits to the exports of arms to third countries considered as human rights perpetrators (Haspeslagh, M., 2011) (Duquet, 2014, p: 40-52).

The explanatory reasons explaining why Dutch parliaments approved the arms exports to third countries considered as human rights perpetrators

In the Netherlands, the public value protecting human rights is very important for the Dutch citizens. This public value is rooted in the Dutch societies from different periods (Kortmann, 2005, p: 22 cited in Mak, 2007, p: 24) as it has been recognized in the Dutch Constitution since the first time it was established in 1815 (Mak, 2007, p: 24). The incorporation of this public value in the Constitution even is meant to protect the Dutch citizens' human rights against the country of the Netherlands. This all and all shows how much this value is important and rooted in the society of the Netherlands (Acharya, S., 2013)(Mak, 2007, p: 25).

Because each parliamentary member is interested to seek re-election, each of them would take any chances to earn more citizens' voices in the next election period, including assuring the governments' respect on the citizens' public values like the public value protecting human rights. The fact that this public value is very important for the Dutch citizens motivates parliaments to stringent their control on the exports of arms to third countries considered as human rights perpetrators, including Indonesia (see Milner, 1997, 35-36 cited in Tacq, R., 2009, p: 28-32) (Mak, 2007) (Tweede Kamer der

Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

Due to the importance of this public value in the Netherlands and the fact that Indonesia has been argued as a human rights perpetrator, one might argue that the exports of corvettes to Indonesia should also have not been allowed by Dutch parliaments. However it is also necessary to look on the possibility that parliaments might have approved those exports because they have been interested to support the governments' interest to gain some economic benefits like providing jobs which is important for the economic growth. Governments' regulations, laws, policies, measures, and decisions, including the decision to grant the arms export permits, influence the growth of arm industries and determine whether the governments' spending in sustaining arm industries has been cost-effective. Countries that do not decide to compete in the international arm trade may still need (for their defense reason) to subsidize the research and development programs in their arm industries to maintain their capability in producing arms. These countries may have not justified that their spending has been cost-effective as their subsidy budget might have been higher than their arm-selling revenues. These countries use tax income to pay this different. This situation may produce budget deficits for other government programs, including education, health care, housing, and infrastructure. Economically justified government regulations, laws, policies, measures and decisions, including the decision to grant the arms export permits, help government prevent this situation to happen (see Brink, T., T., 2014) (Bitzinger, R., A., 2009) (see Woolcock, M., 1998).

Due to the consideration that the economic consequence of each permit decision might be interesting for parliaments and that human rights protection is important for the Dutch society, this thesis investigates the explanatory reasons that motivate parliaments approve the governments' attempts to grant the permits for the arms exports to third countries considered as human rights perpetrators like Indonesia (see Schmelzle, 2011, p: 14) (see Pfaff, 2011, p: 4) (see Schmelzle, 2011, p: 10-11) (Tweede Kamer der Staten-.Generaal, DVB/NW-040/12, 2012, Beantwoording vragen van het lid El Fassed over gebrek aan openheid over mogelijke tankdeal met Indonesie) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië).

THE ROLE OF INTEREST GROUPS IN FRANCE, GERMANY, BELGIUM, AND THE NETHERLANDS IN INFLUENCING THE PARLIAMENT'S APPROVAL

The decision of parliaments to disapprove or to approve the governments' attempt to grant the permit is influenced by the information the interest groups have submitted. This information is based on the citizens' wishes and the information that they have received from their networks. They combine this information to create a good lobby material (see Gilens and Page, 2014) (see Beyers, J., and Kerremans, B., n.d.). The quality of this information is influenced by how good their networks are. These interest groups include the business organizations, the NGOs and the labor unions (see Beyers, J., and Kerremans, B., n.d.).

In general German and French interest groups are more active in building transnational networks compared to Dutch and Belgian groups. Dutch interest groups are however the least Europeanized ones. Dutch interest groups have the lowest contacts with interest groups from other EU Member States and in third countries compared to French, German, and Belgian interest groups. French interest groups are the most adapted ones in seeking contacts and building relationships with others. (see Beyers, J., and Kerremans, B., n.d.).

Due to the minimum networks of Dutch interest groups, they seem to have minimum information on arms business. This minimum information might make parliament become less motivated to grant their approval. This minimum information can also lead to misjudgment on the information about the most recent development of the human rights protection in recipient countries and on the competitors willing to supply arms to third countries whose permit applications have been rejected. For the case of the tanks deal with Indonesia, the unawareness of parliaments that Germany would supply the tanks when the permit would not be granted might be due to this minimum information (see Beyers, J., and Kerremans, B., n.d.).

This thesis considers the Netherlands is an interesting Member State to research in regards to the purpose of this study because it has the complexity explained above. This is interesting for those who want to understand how arms control is done at the national level in the EU and how such control is influenced by the Member States' obligation to implement the ECC's criteria during the control, influenced by the national parliaments' preferences, and influenced by the consideration that these preferences might be based on minimum information supplied by interest groups (see Beyers, J., and Kerremans, B., n.d.).

CHAPTER 3 RESEARCH METHOD AND SOURCES

The purpose of this study is to understand the national parliamentary control provided on the exports of arms in an EU Member State. This thesis aims to understand how this control is used in a Member State to regulate the exports of arms to third countries considered as human rights perpetrators (see Sagepub, n.d.) (see Thomas, 2003).

To clearly understand the application of this parliamentary control, this thesis uses the qualitative research method applied on a case study. The use of this method makes this thesis focus on text instead on numbers of the case being studied. This thesis chooses to study the parliamentary control in the Netherlands provided on the exports of arms to Indonesia. Such "text" is the derived from the parliamentary and ministerial letters, which later will be further explained in this section (see Sagepub, n.d.) (see Thomas, 2003).

The Netherlands is chosen because in the Netherlands there is indeed a parliamentary control provided on the exports of arms to the third countries considered as human rights perpetrators and because of its respect on human rights. This parliamentary control is done through the giving of the parliamentary approval on the governments' attempt to grant a permit. Parliaments usually only

disapprove the governments' attempt to grant the permit for a particular export of arms when it carries risk of violation of human rights (see Sagepub, n.d.) (see Thomas, 2003).

To clearly understand the application of this Dutch parliamentary control, the control provided on the exports of arms to Indonesia is observed. This thesis investigates the explanatory reasons explaining why Dutch parliaments approved the exports of arms to Indonesia (see Sagepub, n.d.) (see Thomas, 2003).

Indonesia is chosen for two reasons as the recipient country. The first reason is because Indonesia has been considered as a human rights perpetrator. Therefore the Dutch parliaments provide control on the exports of arms to Indonesia. The second reason is because the Dutch parliaments have decided to approve the exports of corvettes to Indonesia, but not the export of tanks. This event is interesting to observe as it helps to understand the explanatory reasons behind the parliaments' approval (see Sagepub, n.d.) (see Thomas, 2003).

To understand the explanatory reasons behind the parliaments' approval, this thesis chooses a research question of, "what are the explanatory reasons that explain why Dutch parliaments decide to approve the governments' attempts to grant the permits to the exports of arms to Indonesia?" (see Sagepub, n.d.) (see Thomas, 2003).

The research result of this thesis shows that the parliamentary control is provided on the exports of arms that carry the risk of violation of human rights. The result also shows that the parliaments' approval is given based on the examination on the technical use of the exported products. Parliaments only approve when the exported products, based on their technical use, will not be used to violate human rights. This explains why parliaments only approved the governments' attempt to grant the permits to the exports of the corvettes to Indonesia, but not the thanks. The governments' attempt to grant the permit to the export of tanks was disapproved because tanks are likely to be used to suppress human rights, when corvettes not (see Sagepub, n.d.) (see Thomas, 2003).

The methodology used in for research in this thesis is the qualitative analysis method. This method contains a qualitative nature and is used to analyze an in-depth case study which is the parliamentary control in the Netherlands provided on the exports of arms to Indonesia. This method is applied during the review on the Ministers' letters submitted to the parliaments. This review is to research the parliamentary control on the exports of arms to Indonesia. This review is to analyze the reasoning behind the parliamentary approval (see Sagepub, n.d., p: 321-322) (see Sagepub, n.d.) (see Thomas, 2003).

This thesis uses the in-depth qualitative analysis method to understand what parliaments "really" thought when making decision to approve and to disapprove the governments' attempts to grant the permits. This method is used to understand the parliaments' thought against particular situation at a particular time. This method is used to analyze qualitatively the text in the letters to understand, based on interpretation, the social experience of parliaments when knowing that their governments want to grant the permit for an export of arms to a perpetrator of human rights. The use of an interpretation in understanding the text is named as the hermeneutic perspective method and is part of the in-depth qualitative analysis method. The use of this method in the theoretical framework of this thesis helps this thesis to make interpretations based on scholarship theories (see Patton 2002, p: 114 cited in Sagepub, n.d., p: 321) (see Van Evera, 1997).

The study in this thesis observes the official letters from the Ministers submitted to parliaments since 1998 until 2013. The year of 1998 is chosen because it was the starting point of the obligation to adopt the ECC's criteria. The year of 2013 is chosen in regards to the need to find the needed and accessible research materials (see Sagepub, n.d.) (see Thomas, 2003).

These letters are published online through the Dutch governments' website (www.Rijksoverheid.nl). From this website these letters were downloaded before being observed. This website provides almost all Ministers' letters containing information on the exports of arms in the Netherlands. The relevant information for the topic studied in this thesis is gathered from these letters (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (see Sagepub, n.d.).

In these letters the Ministers provided answers to the parliamentary questions mentioned in the letters that parliaments had earlier submitted to them. These parliamentary questions include some questions on some arms export issues. Through their letters parliaments ask the reasons why governments want to grant the permits to some exports. In each of these letters, the questions from parliamentary members are displayed before the answers from the Ministers. In each of these letters, a reference number is mentioned. This reference number can be used by the parliamentary members from the following election period to find the Ministers' answers on the questions from earlier parliamentary members. This arrangement is to avoid different parliamentary members ask the same questions. The documentation of these letters provides historically information about the arms business in the Netherlands and in the EU (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (see Sagepub, n.d.).

From this historical information, each parliamentary member can find different information on the exports of arms in the Netherlands and in the EU. This information includes the information on the arms that will be exported and on the arms that have been exported. From here parliaments can understand the status of each arms export. For example, from the letters parliaments can understand whether a particular exporter has submitted an arms export permit prohibition application (*sondage*) or not and what is the result of the assessment on this prohibition application. From such information each parliamentary member can decide whether they want to propose a particular motion to prevent the export of those arms. When the whole parliament at the end decides to adopt the proposed motion before government grants the permit and when the government decides to respect the adopted motion, the permit will not be granted (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (see Sagepub, n.d.).

Through their letters parliaments ask some questions on whether particular arms export applications may violate ECC's criteria. Normally, their questions are only to ensure that governments have assessed each application using the ECC's criteria. Through their letters the governments explain that that they have indeed assessed each application using the ECC's criteria, including its criterion protecting human rights. To earn the approval from parliaments, governments also share the motives behind their attempt to grant the permit. From the information that governments provide in their letters, each parliament member can decide whether they want to propose a motion to prevent a particular arms export (Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d.) (see Sagepub, n.d.) (see Thomas, 2003).

This thesis deploys inductive qualitative data analysis helpful to reduce and summarize a complex data (Thomas, 2003, p: 3). Using this method, this thesis identifies and analyzes qualitatively some important categories in the data gathered from the Ministers' letters (Sagepub, n.d., p: 322). Re-reading these letters many times is helpful to identify the categories related to the parliamentary control in the Netherlands provided on the exports of arms to third countries argued as human rights perpetrators. These categories include the government as the actor who grants the export permit, the parliament as the actor who gives the approval on whether a permit should be granted, the products being exported, the recipient country which is Indonesia, the governments' strategies to earn the parliaments' approval and the parliaments' approval itself. These governments' strategies include the governments' respects on the ECC's criteria and on human rights, and motive sharing (see Sagepub, n.d., p: 322) (see Thomas, 2003, p: 2-3).

From these categories, this thesis identifies and analyzes qualitatively their patterns and their relationships. This identification and analysis help this thesis understand when parliaments grant their approvals on the governments' attempts to grant the permits for the exports of arms to Indonesia. To draw the patterns, this thesis observes firstly to what extent parliaments and governments play an influential role in different years to assure that a particular permit should be granted or not. After that, this thesis observes the pattern of the exports of arms to Indonesia. This pattern gives information about the exports of arms from the Netherlands to Indonesia from 1998 – 2013 that have been approved and disapproved by the parliaments. Furthermore, this thesis also observes the pattern of the strategies that Dutch governments have deployed to earn the parliaments' approval. Moreover, this thesis also observes the pattern of the moments when the parliaments grant their approval on the attempts of their governments to grant the permits (Sagepub, n.d., p: 322) (see Thomas, 2003).

The relationship of these patterns are identified and analyzed to understand for the exports of which products (whether the corvettes or the tanks or both) parliaments grant their approval and which governments' strategy(s) that enables governments to earn the parliaments' approval (see Sagepub, n.d., p: 322) (see Sagepub, n.d.) (see Thomas, 2003).

This thesis identifies and analyzes these categories, their patterns and relationships when gathering the data from the letters (see Stake, 1995, cited in Sagepub, n.d., p: 322). This identification and analysis are done based on the theoretical framework explained in Chapter one and Chapter two (Tavallaei, M., and Talib, M., A., n., d., p: 573-574) (see Sagepub, n.d.) (see Thomas, 2003).

This theoretical framework is created from scholarship theories explained in Chapter two. These scholarship theories are usually used by scholars to explain the parliamentary approval on the implementation of a particular international law in a country (Kumm, 2004) (see Sagepub, n.d.) (see Thomas, 2003). These theoretical framework helps this thesis observe and develop perceptions on certain aspects of the parliamentary control based on scholarship theories and identify and analyze the categories, the patterns and the relationships of these categories based on these perceptions (Tavallaei, M., and Talib, M., A., n., d., p: 573-574) (see Sagepub, n.d.) (see Thomas, 2003).

To observe and to develop perceptions on these aspects, the exports of the corvettes and the tanks to Indonesia and the behaviors of governments and parliaments linked to those exports between the periods of 1998 until 2013 are deeply and thoroughly studied (Denzin & Lincoln 2005, Merriam 1998,

Stake 1995, 2006, Yin, 2003 cited in Tavallaei, M., and Talib, M., A., n., d., p: 574) (see Sagepub, n.d.) (see Thomas, 2003).

The use of this theoretical framework for developing theoretically based perceptions before identifying and analyzing the categories, their patterns, and their relationships makes this identification and analysis are done based on particular theories (see Yin, 2008, p. 28 cited in Tavallaei, M., and Talib, M., A., n., d., p: 574) (see Sagepub, n.d.) (see Thomas, 2003). This is necessary to ensure that the finding of this thesis is based on theoretical perspectives (see Creswell, 2007 cited in Tavallaei, M., and Talib, M., A., n., d., p: 574) (see Sagepub, n.d.) (see Thomas, 2003) which is very important to make the finding become theoretically explainable. The use of theories in the research framework helps this thesis link the theories and the causes and effects of each phenomenon linked to the parliamentary control. This helps this thesis develop theoretically explainable perceptions that are based causal laws (see Van Evera, 1997, p: 7-8) (see Sagepub, n.d.) (see Thomas, 2003). For example, according to the finding of this thesis, there are three explanatory reasons explaining why Dutch parliaments approved the governments' attempts to grant the permits for the exports of corvettes to Indonesia. Parliament approved the governments' attempts to grant the permits because governments had explained that they had implemented the ECC's criteria, including its criterion protecting human rights, when assessing the permit applications and because they had shared the economic motive behind the exports. In addition Dutch parliaments disapproved the export of tanks to Indonesia because they had understood that based on their technical use the exported tanks can be used to violate human rights. Although they also understood that the government had implemented the ECC's criteria, including its criterion protecting human rights, when assessing the permit application and that the export of the tanks could decrease the national expenditures (the economic motive of the export), they still decided to disapprove. This shows that governments need to show their respect on human rights not only by assessing the application using the ECC's criterion protecting human rights, but also by assuring that the exported products, based on their technical use, will not be used to violate human rights in the recipient countries. From here, based on scholarship theories, it can be concluded that, the first explanatory reason explaining why Dutch parliaments decide to approve the attempts of their governments to grant the permit to the export of arms to a third country considered as human rights perpetrator like Indonesia is the respect of their governments on the obligation to implement the ECC's criteria during the application assessment. The second explanatory reason is the resinous respect of governments on human rights. This includes implementing the ECC's criterion protecting human rights during the application assessment and assessing the technical use of the exported products to observe whether the products are likely to be used to violate human rights. The third explanatory reason is the motive sharing. The use of the theories in the research framework helps this thesis produce finding that provide specific theoretically explanations of phenomena linked to the national parliamentary control in the Netherlands provided on the arms exports to third countries considered as human rights perpetrators (see Van Evera, 1997, p: 15-16) (see Sagepub, n.d.) (see Thomas, 2003).

CHAPTER 4 RESEARCH RESULTS

According to the research finding, although according the Dutch law governments have the sole right to decide whether a permit for should be granted or not, when parliaments want to prevent a particular arms export, parliaments can create and adopt a parliamentary motion. A parliamentary motion functions as a "common position" of all parliamentary members against a particular government's attempts, including the attempt to grant the permit for that particular arms export. When government decides to respect the adopted motion, they will not grant the permit (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

One example for this is the motion proposed by the parliamentary member El Fassad and created to gather the voices of other parliamentary members to take a stand against government's attempt to grant the permit to the export of leopard tanks to Indonesia. When there is a motion established against a particular arms export, although the Dutch law does not oblige government to adopt the motion, parliaments require governments to respect that motion and grant the denial permit to the exporter (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de

leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

Because the motions are used by parliaents to influence the governments' permit decision, this thesis argues that, when there is no specific parliamentary motion created against a particular export, there is no attempt from the parliaments to dprevent the governments from granting the permit to any arms exports (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

From the period of 1998 until 2013, there has been only one motion created to prevent an arms export to Indonesia. This motion is the motion proposed by the parliamentary member El Fassad mentioned earlier. This motion was proposed in November 2011 and adopted by the parliament in December 2011. This motion was meant to avoid the export of leopard tanks to Indonesia. In other words, from the period of 1998 until 2013, parliaments have decided to approve other arms exports to Indonesia, including the exports of corvettes. To observe the explanatory reasons explaining why parliaments approved the exports of corvettes and disapproved the exports of tanks, this thesis uses a qualitative analysis based on some theories explained in Chapter two (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p. 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

These theories are based on earlier scholarship studies. As explained in earlier chapters, according to legal scholars, parliaments approve the governments' attempts to grant a permit for the export of arms to third countries considered as human rights perpetrators when those attempts respect the existing law (see Pfaff, 2011, p: 4).

In 2004 the government informed the parliaments through their letter that they wanted to grant a permit to an export of corvettes to Indonesia. Through that letter government also informed parliaments that they had assessed the permit prohibition application (sondage) for this export using the ECC's criteria. On 17 October 2006, through their letter the government informed the parliament once again that they also had assessed on the application for this export using the ECC's criteria (see Pfaff, 2011, p: 4) (Tweede Kamer der Staten-.Generaal, BEB/HIB 4051879, 2004, "Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020)") (Tweede Kamer der Staten-.Generaal, Besluitvorming m.b.t. korvetten voor Indonesië, 2006). Because between 2004 until 2006 there was no parliamentary motion adopted against this export, this thesis argues that one of explanatory reasons explaining why Dutch parliaments approved the export of these corvettes to Indonesia is the fact that governments had implemented the ECC's criteria when assessing these applications. This explanatory reason explains that parliaments will grant their approval on the governments' attempt to grant a permit to an arms export when they know that governments respect their obligation to implement the ECC's criteria during the application assessment (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

However, according to the finding, through their letter, to earn the parliaments' approval on their attempts to grant the permit to the export of tanks to Indonesia in 2012, governments explained to parliaments that they had also implemented the ECC's criteria when assessing the permit application. This means that, although parliaments only approved the exports of corvettes, but not the export of tanks, for the exports of both types of products governments had assessed the applications using the ECC's criteria. From here, this thesis argues that the governments' compliance on their obligation to implement the ECC's criteria when assessing the application only becomes one of the explanatory reasons explaining why parliaments approve the exports of arms to Indonesia when the exported products are corvettes, but not tanks (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p. 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

In addition to this, the scholars also argue that one more explanatory reason to earn parliaments' approval is the governments' respect on the public values of the society. This thesis argues that, one of the important public values for the Dutch society is the public value protecting human rights. In the

Netherlands, the governments' respect on human rights is very important to earn the parliaments' approval on their attempts. From the finding, one of the explanatory reasons explaining why Dutch parliaments approved the export of corvettes to Indonesia might be the fact that governments had informed parliaments that, they had assessed the permit prohibition application and the permit application for this export using the ECC's criterion protecting human rights. Assessing the permit application using the ECC's criterion protecting human rights is one way to inform parliaments that governments have taken into account the need to protect human rights when assessing the applications and to show to parliaments that they care with human rights (Lagerwaard, 2009) (see Schmelzle, 2011, p: 10-11) (Tweede Kamer der Staten-.Generaal, BEB/HIB 4051879, 2004, "Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020)") (Tweede Kamer der Staten-.Generaal, Besluitvorming m.b.t. korvetten voor Indonesië, 2006).

From here, using the scholars' theories this thesis argues that the two explanatory reasons explaining why the parliaments approved the exports of corvettes to Indonesia are the governments' respect on the ECC's criteria and the governments' respect on human rights (Tweede Kamer der Staten-.Generaal, Besluitvorming m.b.t. korvetten voor Indonesië, 2006) (Tweede Kamer der Staten-Generaal, Antwoorden op Kamervragen over de mogelijke leverantie van korvetten aan de Indonesische marine (2030418020), 2004) (Lagerwaard, 2009).

According to scholars, another source to earn the parliaments' approval is the motive sharing. As explained in earlier parts of this thesis, some scholars also argue that parliaments will approve an export of arms when governments share the (economic) motives behind the export. This thesis observes whether one of the explanatory reasons explaining why Dutch parliaments approve the exports of arms to Indonesia is the fact that governments always share the motives behind those exports (see Schmelzle, 2011, p: 14).

According to the finding, Dutch governments always inform their parliaments about the economic consequences of every permit decision. For example, through their letter on March 2005 government shared the motive behind their attempt to grant the permit to the exports of corvettes to Indonesia. They explained that the one of those motives is the economic motive. They explained that refusing to grant the permits to the exports of corvettes would make the Netherlands pay the insurance loss to the banks funding the production of those corvettes. From here, this thesis argues that one of the explanatory reasons explaining why parliaments approved the exports of arms to Indonesia is the governments' motive sharing (Tweede Kamer der Staten-Generaal, Reactie op schriftelijke vragen vanuit de vaste commissie voor Economische Zaken, 2005).

However, this thesis also finds out that, in the Netherlands, motive sharing and showing the respect on the ECC's criteria and human rights by assessing the applications using the ECC's criteria, including its criterion protecting human rights, do not always motivate parliaments to grant their approval. One example of this is what happened in 2012 when parliaments refused to grant their approval on the export of tanks to Indonesia. To prevent this tanks export parliament adopted the motion proposed by El Fassad in 2011. After this motion had been adopted, in 2012, to share the motives behind this export, government explained that the economy benefit behind this tank export would be about € 200 million. In addition, government also explained that these tanks had been the surplus tanks of the Dutch Defense department, therefore their export would not only provide export revenue, but

also would decrease the annual national expenditures. To show their respect on the ECC's criteria and human rights, government also explained that they had assessed the permit application for this export using the ECC's criteria, including its criterion protecting human rights. Despite these explanations, at the end parliaments still decided to deploy the motion from El Fassad to prevent government from granting the permit to this export. This phenomenon shows that motive sharing and showing the respect on the ECC's criteria and human rights by assessing the applications using the ECC's criteria, including its criterion protecting human rights, are not always being the reasons explaining why Dutch parliaments grant their approval on the exports of arms to third countries considered as human rights perpetrators (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesieë).

Moreover in June 2012, through another letter government explained their attempt to grant a permit for another export of corvette to Indonesia. In this letter, to inform parliament about the legal background of their attempt and their respect on human rights, government explained that the assessment using the ECC's criteria, including its criterion protecting human rights, on the permit prohibition application for this export had produced a positive result. In addition, to share the economic motive behind this export, government explained that its value is \$ 220 million (approximately € 175 million) and it provides about 375 men of work. Since the letter was submitted to parliament until the time this thesis is being produced, there has not been any motion adopted by parliaments to prevent this export. In other words, until this thesis is being produced, parliaments have not had any attempt to prevent this export (Tweede Kamer der Staten-.Generaal, DVB/NW-405/12, 2012, Beantwoording vragen van de leden Ten Broeke en Schaart over de levering van een SIGMA korvet aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

Although for the exports of corvettes and tanks to Indonesia, governments had assessed the applications using the ECC's criteria, including its criterion protecting human rights, and had shared the motives behind the exports; parliaments only approved the exports of corvettes, but not the export of tanks. The finding shows that the reason behind these two different decisions is the fact that the assessment on each permit application involves the assessment on the likeliness that the exported products will be used to violate human rights, which is most of the time evaluated based on their technical use. Parliaments approved the exports of corvettes because, based on their technical use, it is not likely that the exported corvettes would be used to violate human rights in Indonesia. On the other hand because based on their technical use it is likely that the exported tanks will be used to violate human rights in Indonesia, parliaments disapproved the export of these tanks. This all and all shows that governments need to show their respect on human rights not only by assessing the

application using the ECC's criterion protecting human rights, but also by assuring that the exported products, based on their technical use, will not be used to violate human rights in recipient countries (Tweede Kamer der Staten-.Generaal, DVB/NW-405/12, 2012, Beantwoording vragen van de leden Ten Broeke en Schaart over de levering van een SIGMA korvet aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-332/12, 2012, Beantwoording vragen van de leden Van Dijk en Van Bommel over de verkoop van tanks aan Indonesieë) (Tweede Kamer der Staten-.Generaal, BS 2012012493, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië) (Tweede Kamer der Staten-.Generaal, DVB/NW-507/12, 2012, Verkoop van Leopard 2A6 tanks met toebehoren aan Indonesië, Beantwoording vragen van het lid El Fassed over berichten dat Indonesië mogelijk van Duitsland Leopard- tanks wil kopen) (Akkerman, 2012, p: 12) (Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen") (Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?").

CHAPTER 5 CONCLUSION

This thesis has found at least three reasons explaining why national parliaments of an EU Member State respecting human rights like the Netherlands approve the attempts of their governments to grant the permits to the exports of arms to third countries considered as human rights perpetrators like Indonesia. The first explanatory reason is the compliance of the government with the existing law, including the obligation to implement the ECC's criteria during the application assessment. The second explanatory reason is the seriousness of governments to care with human rights. Governments are not only expected to implement the ECC's criterion protecting human rights during the application assessment, but also assessing the technical use of the exported products to observe whether the products are likely to be used to violate human rights in recipient countries. Dutch parliaments will only approve an arms export to a third country considered as human rights perpetrator like Indonesia when they know that the exported arms, based on their technical use, will not be used to violate human rights in recipient countries. The third explanatory reason is the (economic) motive behind the export.

Among these three explanatory reasons, this thesis has found out that, in an EU Member State respecting human rights like the Netherlands, the governments' respect on human rights is the most important source to earn the parliaments' approval on an export of arms to a third country considered as human rights perpetrator like Indonesia. For the Member States respecting human rights like the Netherlands, the concern to prevent violation of human rights in recipient countries is more important than the concern to earn economic benefit from an export. Although Dutch parliaments had understood that the export of tanks would give not only revenue but also reducing national expenditures, as the tanks had been the surplus tanks from Defense department, they still decided to disapprove the export. This all and all shows that the arms export control in different EU Member States after the establishment of the ECC's criteria is not the same. This is because the

national parliaments in the Member States that have more respect to human rights, like the Netherlands, may prevent their governments from granting the permits to the exports that carry risk of violation of human rights although they have known that their governments have assessed the permit applications using the ECC's criterion advocating the need to protect human rights. On the other hand, the national parliaments in the Member States that have less respect on human rights, like France, Germany, and Belgium, may still allow their governments to grant the permits for such exports, because for them export revenues, bilateral relationship and the need to compete might be more important than human rights. This thesis argues that to find out which one of these explanatory reasons that is more important than the rest to earn the approval from the national parliaments in each of these Member States might be an interesting topic for the following research as it gives clear picture about the national preferences in other Member States that influence their arms export controls.

REFERENCE

Acharya, S., 2013, "National Human Rights Institutions (NHRIs) and their role in the Policy Making Process: A Look at the Impact of the Equality and Human Rights Commission (EHRC) in the United Kingdom in the Context of Diminishing Power", available at:

http://munin.uit.no/bitstream/handle/10037/5459/thesis.pdf?sequence=1

Akkerman, M., 2012, "Hoge kwaliteit, zo goed als nieuw": tweedehands wapens uit Nederland", available at: http://www.stopwapenhandel.org/sites/stopwapenhandel.org/files/zogoedalsnieuw.pdf

Archive.org, n. d., OPMERKINGEN OVER DE WET HOUDENDE VERBODSBEPALINGEN TEGEN HET DRAGEN VAN WAPENEN, available at:

https://archive.org/stream/opmerkingenover00elstgoog/opmerkingenover00elstgoog_djvu.txt

Bailes, A., Depauw, S., and Baum, T., 2011, "The EU defense market: balancing effectiveness with responsibility", Drukkerij Artoos

Balzacq, T., 2011, 'Securitization Theory, How Security Problems Emerge and Dissolve', Routledge, New York and London.

Bartlett, J., Miller, C., 2010, "The Power of Unreason Conspiracy Theories, and Extremism, and Counter Terrorism", Demos.

Beyer, 2009, 'The Same or Not the Same – On the Variety Mechanisms of Path Dependence', International Journal of Human and Social Sciences.

Bilge, L., and Dumitras, T., 2012, "Before We Knew It, An Empirical Study of Zero-Day Attacks In The Real World", Releigh, North Carolina, USA.

Bromley, M., 2008, "The Impact on Domestic Policy of the EU Code of Conduct on Arms Exports, The Czech Republic, the Netherlands and Spain", SIPRI Policy Paper No. 21, available at: http://books.sipri.org/files/PP/SIPRIPP21.pdf.

Bromley, M., 2012, "The review of the EU common position on arms exports: prospects for strengthened controls", available at:

http://www.sipri.org/research/disarmament/eu-consortium/publications/publications/non-proliferation-paper-7

Buzan, B., Hansen, L., 2009, "The Evolution of International Security Studies", Cambridge University Press, UK.

Centrale Archief Selectiedienst, Nationaal Archief, Den Haag 1995, Nummer archiefinventaris: 2.06.088, "Inventaris van het archief van de Centrale Dienst voor In- en Uitvoer, 1941-1964", available at: file:///C:/Users/hp/Downloads/NL-HaNA_2.06.088.ead.pdf

Centrale Archief Selectiedienst, Nationaal Archief, Den Haag 2006, Nummer archiefinventaris: 2.06.107, "Inventaris van het archief van het Ministerie van Economische Zaken: Directoraat-Generaal voor de Buitenlandse Economische Betrekkingen, 1945-1976", available at: file:///C:/Users/hp/Downloads/NL-HaNA_2.06.107.ead%20(1).pdf

Christiansson, M., 2010, "Solidarity and Sovereignty – The Two Dimensional Game of Swedish Security Policy", Baltic Security and Defense Review, Volume 12, issue 2, 2010.

Connection and Constitution: Locating War and Culture in Globalization Studies, available at: http://www.international.ucla.edu/media/files/44.pdf?AspxAutoDetectCookieSupport=1

Constructivism and Securitization Studies, available at:

http://graduateinstitute.ch/files/live/sites/iheid/files/sites/developpement/shared/developpement/cours/E777/Securitization_Balzacq.pdf

Cooper, N., 2012, "The Arms Trade Treaty in the Context of Post-Cold War Conventional Arms Trade Regulation", The Trust for Research and Education on the Arms Trade and the British Academy.

Council of The European Union – User's Guide, 2009, "User's Guide to Council Common Position 2008/944/CFSP defining common rules governing the control of exports of military technology and equipment".

COUNCIL REGULATION (EC) No 428/2009, The Council of the European Union, 2009. Available at: http://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=OJ:L:2009:134:0001:0269:en:PDF

COUNCIL COMMON POSITION 2008/944/CFSP, The Council of the European Union, 2008. Available at: http://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=OJ:L:2008:335:0099:0099:EN:PDF

DIRECTIVE 2009/43/EC, The European Parliament and the Council, 2009. Available at: http://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=CONSLEG:2009L0043:20101214:EN:PDF

Depauw, 2010, "The Common Position on arms exports in the light of the emerging European defense market", Brussel

De Graaf, B., 2011, "Evaluating Counterterrorism Performance", A comparative study, Routledge, London and New York.

Detjen, J., n.d., "Core Values of the German Basic Law: A Source of Core Concepts of Civic Education", available at: http://www.civiced.org/pdfs/GermanAmericanConf2009/Paper_JDetjen.pdf

Diskaya, A., 2013, "Towards a Critical Securitization Theory: The Copenhagen and Aberyswyth Schools of Security Studies", e-International Relations, available at:

http://www.e-ir.info/2013/02/01/towards-a-critical-securitization-theory-the-copenhagen-and-abery stwyth-schools-of-security-studies/#_ftn7 access time: 15/03/13, 09:35am - 21:56pm.

Dodd, D., Lyklema, M., and Van Weringh, K., D., 2006, "A Cultural Component as an integral part of the EU's Foreign Policy?", available at:

 $\frac{http://static1.squarespace.com/static/526e5978e4b0b83086a1fede/t/546f49a2e4b0a19016cd31c1/1416579490067/Introduction%2C+summary%2C+conclusions.pdf}{}$

Drgoncova, J., 2011, "THE FREEDOM OF RELIGION OR BELIEF", available at: http://arno.uvt.nl/show.cgi?fid=114882

Duke, S., 2000, 'The Elusive Quest for European Security, from EDC to CFSP', Antony Rowe Ltd., UK.

Duquet, N., 2014, "Business as usual?, Assessing the impact of the Arab Spring on European arms export control policies",

http://www.flemishpeaceinstitute.eu/sites/vlaamsvredesinstituut.eu/files/report_business_as_usual _web.pdf

Eisbach, J., and Rilling, R., n. d., "SDI en de beperkte spin-off van militaire R&D", available at: http://www.tpedigitaal.nl/assets/static/9_2_Dec_85/SDI_en_de_beperkte_spin_off_van_militaire_R D_Joachim_Eisbac.pdf

Ericsson, L., 2008, The Swedish Arms Trade and the Politics of Human Rights: A Comparative Case-study of Swedish Weapon Exports to the Islamic Republic of Pakistan and the Russian Federation in the year of 2006, available at: http://www.diva-portal.org/smash/get/diva2:3582/FULLTEXT01.pdf

European Commission, 2013, "News, EU internal defense market is opening slowly". Available at: http://ec.europa.eu/enterprise/newsroom/cf/itemdetail.cfm?item_id=6037&lang=en (last access on 23/03/2013 at 12:43 p.m.)

"Faithfull Allies?", n.d., "Faithfull Allies? Dutch reactions to the American Vision of German Rearmament, 1945-1955", available at:

http://dspace.library.uu.nl/bitstream/handle/1874/237690/Faithfull%20Thesis%20YWE%20Tadema%200474711%20.pdf?sequence=1

Foster, N., 2012, "Blackstone's EU Treaties & Legislation 2012-2013", 23rd edition, Oxford University Press.

Fu, P., P., Kennedy, J., Tata, J., Yukl, G., Bond, M., H., Peng, T., Srinivas, E., Howell, J., P., Prieto, L., Koopman, P., Boonstra, J., J., Pasa, S., Lacassagne, M., Higashide, H., and Cheosakul, A., 2004, "The impact of societal cultural values and individual social beliefs on the perceived effectiveness of

managerial influence strategies: a meso approach", Journal of International Business Studies 35, p: 284–305.

Furedi, F., 2008, "Fear and Security: A Vulnerability-led Policy Response", Social Policy and Administration, Vol. 42, No. 6.

Gahetna.nl, n.d., Archieven - Zoeken: Stukken betreffende bevordering van de uitvoer.1949-1963 periode van 1949 tot 1963, available at:

http://www.gahetna.nl/collectie/archief/index/index/zoekterm/Stukken%20betreffende %20bevordering%20van%20de%20uitvoer.%201949-1963/beginjaar/1949/eindjaar/1963/aantal/20

Gilens, M., and Page, B., I., 2014, "Testing Theories of American Politics: Elites, Interest Groups, and Average Citizens", available at:

http://scholar.princeton.edu/sites/default/files/mgilens/files/gilens_and_page_2014_-testing_theories_of_american_politics.doc.pdf

Gyarmati, I., Wikler, T., Remillard, M., Vesel, S., 2002, "Post-Cold War Defense Reform, Lessons Learned in Europe and the United States", Potomac Books, Inc., USA. Available at: http://books.google.nl/books?

id=G32Hf131kAcC&pg=PA316&lpg=PA316&dq=sweden+defense+after+cold+war&source=bl&ots=PD Cn1xMp1K&sig=-_E4v5pEbolnEY0qHhuAdcRea5c&hl=nl&sa=X&ei=UnJUUYuFl83y0gWx2YGQCQ&ved =0CC8Q6AEwATgK#v=onepage&q=sweden%20defense%20after%20cold%20war&f=false

Graham, A., Seldon, A., 1990, 'Government and Economies in the postwar world, economic policies and comparative performance, 1945-1985', Routledge, London, UK.

Grebe, J., 2014, "Courage to change or continuation of structural deficits?", available at: https://www.bicc.de/uploads/tx_bicctools/focus13_e.pdf

Gustavsson, J., 1998, 'The Politics of Foreign Policy Change, Explaining the Swedish Reorientation on EC Membership'.

Handboek Douane (HD), 14.80.00 Formulier 302, "302 voor Extern Douanevervoer", available at: http://www.belastingdienst.nl/bibliotheek/handboeken/html/boeken/HD/formulier_302-inleiding_302_voor_extern_douanevervoer.html

Hansen, L., Waever, O., 2002, 'European Integration and National Identity: The challenge of the Nordic states', Routledge, London and New York, USA and UK.

Hansen, T., and March, N., 2013, "Normative power and organized hypocrisy: European Union member states' arms export to Libya", available at:

http://www.tandfonline.com/doi/full/10.1080/09662839.2014.967763#.VJ2shlIRAhttp://www.tandfonline.com/doi/full/10.1080/09662839.2014.967763%23.VJ2shlIRA

Haspeslagh, M., 2011, "THE BELGIAN BURQA-BAN", available at: http://lib.ugent.be/fulltxt/RUG01/001/892/123/RUG01-001892123 2012 0001 AC.pdf

Holland, M., 2005, 'Common Foreign and Security Policy', 2nd Edition, Continuum International Publishing Group, UK.

Hunter, R., 2002, "The European Security and Defense Policy, NATO's Companion or Competitor?", RAND Europe.

In – en uitvoer de afwijzing van de aanvragen tot invoer van 50 000 dozijn paar Italiaanse dames-kousen, available at: http://www.statengeneraaldigitaal.nl/document/tekst?id=sgd %3A19581959%3A0001239&pagina=1

In- en uitvoerbesluit industriële goederen, liigl, n.d., available at: https://www.liigl.nl/Search/? q=In-+en+uitvoerbesluit+industri%C3%ABle+goederen&pg=13

INTERNATIONALE RECHTSHANDHAVING, available at:

http://www.clingendael.nl/sites/default/files/20070000_cscp_chapter_leurdijk.pdf

Jancos, A., Reynolds, C., 2007, "The Political and Security Committee: Governing in the Shadow", European Foreign Affairs Review 12, Kluwer Law International BV, the Netherlands.

Jensen, M., D., and Martinsen, D., S., 2014, "Out of Time? National Parliaments and Early Decision-Making in the European Union", available at:

http://polsci.ku.dk/english/research/healthcare regulation in the european union/publications/M artinsen Dagnis Jensen.pdf

Jordan, R., 2001, "Library of Congress Cataloging-in-Publication Data", Greenwood Publishing Group Inc., USA.

Kamerstuk 22054 Wapenexportbeleid, 2013, available at: available at: https://zoek.officielebekendmakingen.nl/kst-22054-223.html

Kampen, R., 2013, "De Effectiviteit van het Arms Trade Treaty", available at: file:///C:/Users/hp/Downloads/Scriptie%20(1).pdf

KONINKLIJKE BOODSCHAP, "Wijziging en aanvulling van de Vuurwapenwet 1919", available at: http://statengeneraaldigitaal.nl/document/tekst?id=sgd%3A19311932%3A0001085&pagina=1 '

KONINKLIJKE BOODSCHAP, "wet houdende nadere voorzieningen op het stuk van vuurwapenen en munitie", available at: http://resourcessgd.kb.nl/SGD/19181919/PDF/SGD_19181919_0002034.pdf

Kumm, M., 2004, The Legitimacy of International Law: A Constitutionalist Framework of Analysis, http://eiil.org/pdfs/15/5/397.pdf

Lak, M., 2011, "BECAUSE WE NEED THEM..., German-Dutch relations after the occupation: economic inevitability and political acceptance, 1945-1957", available at: http://www.eshcc.eur.nl/fileadmin/ASSETS/eshcc/chr/Thesis_Lak.pdf

Lantis, J., S., 2006, "Strategic Culture: From Clausewitz to Constructivism", Prepared for Defense Threat Reduction Agency.

Laursen, F., 1997, ""The EU Neutrals, CFSP, and Defense Policy", The Thorkil Kristensen Instituut, South Jutland University Centre, Denmark.

Long, D., 1995, "The CFSP and beyond, The EU's Territorial and Functional Conceptions of Security", Carleton University.

Lord, C., 2000, "Legitimacy, Democracy and the EU: when abstract questions become practical policy problems", Policy Paper, University of Leeds.

Love, J., O'Brien, J., 2003, 'Western Europe 2003, Regional Surveys of the World', 5th Edition, Europa Publication Limited 2002, London, UK.

MAXWELL SCHOOL OF CITIZENSHIP AND PUBLIC AFFAIRS SYRACUSE UNIVERSITY, available at: http://www.eliamep.gr/old/eliamep/files/05.01.pdf

Marácz, L., 2009, "Statement by the European Committee Hungarian Human Rights Central Europa Foundation Amsterdam on the Case of the Human Rights Violations of European Citizens in Bolivia", European Committee Human Rights Hungarians Central Europe.

Mahoney, J., 2001, 'Path-Dependence Explanations of Regime Change: Central America in Comparative Perspective, Studies in Comparative International Development, Vol. 36, No. 1., p: 111-141.

Mahoney, J., and Schensul, D., 2008, 'Historical Context and Path Dependence', cited in, 'The Oxford Handbook of Contextual Political Analysis', Chapter 24, Edited by Goodin, R., and Tilly, C., University Oxford Press Inc., New York, USA

Mak, C., 2007, p: 24, "Fundamental Rights in European Contract Law", available at: http://dare.uva.nl/document/2/52678

Matlary J.,H., and Osterud, O., 2007, "Denationalization of Defense, Convergence and Diversity", Ashgate Publishing Limited, England, USA. Available at: http://books.google.nl/books? id=D0SNK03KqFAC&pg=PA136&lpg=PA136&dq=sweden+defense+bill+1995+1996&source=bl&ots=lY sxOFbheh&sig=NkQGOLXd53JKKnX2WyDd746LTjM&hl=nl&sa=X&ei=WfNSUd6-N9Kk0AXO4oGQDg&v ed=0CEcQ6AEwBA#v=onepage&q=sweden%20defense%20bill%201995%201996&f=false

Meyer, C., O., 2004, "Theorizing European Strategic Culture, Between Convergence and the Persistence of National Diversity", Center for European Policy Studies, CEPS Working Document No. 204.

Michel, Q., Tsukanova, M., 2011, "The European Union Export Control Regime of Arms: Comment of the Legislation: article-by-article", Liege University, Belgium. Available at: http://local.droit.ulg.ac.be/jcms/service/file/20120328135409_Vademecum-weapons-01.pdf

Miller, G., 2000, 'Rational Choice and Dysfunctional Institution, Governance: An International Journal of Policy and Administration', Vol. 13, no. 4, Blackwell Publisher, USA

Minister van Landbouw, 2006, Natuur en Voedselkwaliteit, "Vaststelling selectielijst neerslag handelingen Minister van Landbouw, Natuur en Voedselkwaliteit beleidsterrein In- en uitvoerregelingen vanaf 1945", available at:

file:///C:/Users/hp/Downloads/stcrt-2006-191-p7-SC76854%20(1).pdf

Minister van Landbouw, 2006, Natuur en Voedselkwaliteit, "Vaststellingsbesluit selectielijst neerslag handelingen beleidsterrein In- en uitvoerregelingen vanaf 1945 (Minister van Landbouw, Natuur en Voedselkwaliteit)", available at:

http://wetten.overheid.nl/BWBR0020172/geldigheidsdatum_17-12-2013

Minister van Veiligheid en Justitie, n.d., "Regeling wapens en munitie-geldt op dec 2014", available at: http://wetten.overheid.nl/BWBR0008800/geldigheidsdatum_13-12-2014#1

Minister van Veiligheid en Justitie, n.d., "Circulaire Wapens en Munitie 2013 (CWM 2013)", available at: http://wetten.overheid.nl/BWBR0032774/geldigheidsdatum_16-01-2014#A_1

Minister van Veiligheid en Justitie, 1996, "Regeling wapens en munitie-geldt op 1996", available at: http://www.svdetreffers.nl/documents/wwm/wwm-regeling.pdf

Minister van Veiligheid en Justitie, 1996, "Vaststellingsbesluit selectielijst neerslag handelingen beleidsterrein Wapens en Munitie periode 1945–1997",

http://wetten.overheid.nl/BWBR0018753/volledig/geldigheidsdatum_13-12-2014

Minister van Veiligheid en Justitie, 1997, "Vervallen na de inwerkingtreding van Staatsblad 579/1995, per 1 januari 1997, wet van 16 november 1995 tot herziening van de Wet wapens en munitie", available at: http://svprinshendrik.nl/downloads/wwm.pdf

Mousnier, R., 2004, 'The Stability of Europe, The Common Markets towards the European Integration of Industrial and Financial Markets? (1958-1968)', Centre Roland-Mousnier, Paris, France.

Nordstorm, B., 2000, 'Scandinavia since 1500', University of Minnesota, USA.

OECD, 1999, "European Principles for Public Administration", SIGMA Papers, No. 27, OECD Publishing, available at: http://dx.doi.org/10.1787/5kml60zwdr7h-en and at: http://www.oecd-ilibrary.org/docserver/download/5kml60zwdr7h.pdf? expires=1420577374&id=id&accname=guest&checksum=5775C56BBDF74362D5D31B13AACF68E4

Pandey, S., K., and Dixit, P., K., 2011, "The Influence of Culture on Consumer Behaviour", VSRD International Journal of Business and Management Research.

Parlement.com, Moet een motie altijd worden uitgevoerd?, n.d., "Motie een door de Kamer aangenomen motie altijd worden uitgevoerd?", available at:

http://www.parlement.com/id/vhnnmt7mkvvx/moet een motie altijd worden uitgevoerd

Pauly Junior, R., J., 2013, "The Ashgate Research Companion to US Foreign Policy", available at: https://books.google.nl/books?

id=PNsoijrQm1UC&pg=RA2-PA43&dq=mimic+foreign+policies&hl=nl&sa=X&ei=tmS2VNq5DIG3UJWO hLgE&ved=0CCIQ6AEwAA#v=onepage&q=mimic%20foreign%20policies&f=false

Peters, S., and Westphal, K., 2004, "The CFSP/ESDP: from the Tail Light to the Future Motor of European Integration?", Module for a Study Guide on European Integration, Deakin University Australia.

Pfaff, F., 2011, "Enhancing the Legitimacy of the EU, The Contribution of the German Parliament", available at: http://arno.uvt.nl/show.cgi?fid=132434.

Pierson, 2000, 'The Limits of Design: Explaining Institutional Origins and Change, Governance: An International Journal of Policy and Administration, Vol. 13, No. 4, October 2000 (p: 475-499), Blackwell Publishers, USA.

Poullie, Y., D., 2014, "Arms for despots and the powerlessness of public opinion", available at: https://tampub.uta.fi/bitstream/handle/10024/96574/gradu07219.pdf?sequence=1

Provan, K., G., and Milward, H., B., 2001, "Do Networks Really Work? A Framework For Evaluating Public-Sector Organizational Networks", Public Administration Review, Vol. 61, No. 4 (Jul. - Aug., 2001), pp. 414-423

Reichenbach, T., 2008, 'A Brief Survey on Normative Power Europe'

Rijksoverheid.nl, available at:

http://www.rijksoverheid.nl/documenten-en-publicaties/kamerstukken?

keyword=wapenexport&period-from=&period-to=&department=&informationtype=kamerstukken&page=1

Rood, E., 2010, "Het toerekenen van een intracommunautaire transactie binnen een ketentransactie", available at: http://arno.uvt.nl/show.cgi?fid=107509

Sagepub, n.d., "Qualitative Data Analysis", Chapter 10, available at: http://www.sagepub.com/upm-data/43454 10.pdf

Soetendorp, B., 2014, "Foreign Policy in the European Union: History, theory & practice", available at: https://books.google.nl/books?

Staatsblad 1962, 295, Enforcing European Union Law on Exports of Dual-use Goods, available at: https://books.google.nl/books?

id=QGnK91AtpJ4C&pg=PA118&lpg=PA118&dq=Staatsblad+1962,+295&source=bl&ots=5BNktVi5IN&sig=Wvb11Z2jNOhIK5QzfHR-KdB0kns&hl=nl&sa=X&ei=5giPVNajFM6v7AajplCgAg&ved=0CC8Q6AEwAw#v=onepage&q=Staatsblad%201962%2C%20295&f=false

Schmelzle, C., 2011, "Evaluating Governance. Effectiveness and Legitimacy in Areas of Limited Statehood".

http://www.sfb-governance.de/en/publikationen/working_papers/wp26/SFB-Governance-Working-Paper-26.pdf.

Schmidt, I., and Evans, B., 2012, 'Social Democracy After the Cold War', AU Press, Athabasca University.

Security Practices, available at: http://www.didierbigo.com/documents/SecurityPractices2010.pdf

Seidelman, R., n.d., "Problems and Prospects of the Common Foreign and Security Policy (CFSP) and European Security and Defense Policy (CESDP): A German View.

Smith, K., 2000, 'The End of Civilian Power EU: A Welcome Demise or Cause for Concern?', The International Spectator volume XXX5.

Stone, D., 2002, "Policy Paradox, The Art of Political Decision Making", New York: W. W. Norton & Company. Revised edition.

Tacq, R., 2009, "Arms Export Regulations & Member State Preferences".

Tavallaei, M., and Talib, M., A., n., d., p: 573-574, "A General Perspective on Role of Theory in Qualitative Research", available at:

http://www.sosyalarastirmalar.com/cilt3/sayi11pdf/tavallaei_abutalib.pdf

Tweede Kamer der Staten-Generaal, Zitting 1974-1975, Verlenging van de werking van krachtens de In- en uitvoerwet (Stb. 1962, 295) ten aanzien van de In- en uitvoer van goederen gestelde regelen, available at: http://resourcessgd.kb.nl/SGD/19741975/PDF/SGD 19741975 0003959.pdf

Tweede Kamer der Staten Generaal, de motie-El Fassed c.s. over niet leveren van tanks aan Indonesië (33000-X, nr. 47), n.d., "Handelingen", available at:

 $https://zoek.officielebekendmakingen.nl/handelingen/TK/2011-2012/35/h-tk-20112012-35-12? \\ resultIndex=11\&sorttype=1\&sortorder=4$

Tweede Kamer der Staten-Generaal, Vergaderjaar 1985-1986, 19 506, Verlenging van de werking van krachtens de In-en uitvoerwet (Stb. 1962, 295) ten aanzien van de uitvoer van bepaalde staalprodukten gestelde regelen, available at:

http://resourcessgd.kb.nl/SGD/19851986/PDF/SGD_19851986_0007689.pdf

Tweede Kamer der Staten-Generaal, Vergaderjaar 1987-1988, 10 272, Verlenging van de werking van krachtens de In- en uitvoerwet (Stb. 1962, 295) ten aanzien van de in- en uitvoer van goederen gestelde regelen, available at:

http://resourcessgd.kb.nl/SGD/19871988/PDF/SGD_19871988_0005177.pdf

Tweede Kamer der Staten-Generaal, 24 107, nr. 5 NOTA NAAR AANLEIDING VAN HET VERSLAG, "Herziening van de Wet wapens en munitie", available at:

https://zoek.officielebekendmakingen.nl/dossier/24107/kst-24107-5? resultIndex=6&sorttype=1&sortorder=4

Tweede Kamer der Staten-Generaal, n. d., "Wijziging van de Wet werk en bijstand en enkele andere sociale zekerheidswetten (Wet maatregelen Wet werk en bijstand en enkele andere wetten)", available at:

file:///C:/Users/hp/Downloads/memorie-van-antwoord-wetsvoorstel-maatregelen-wet-werk-en-bijst and-en-enkele-andere-wetten.pdf

Tweede Kamer der Staten-Generaal, antwoorden op Kamervragen Lid Algra inzake Irak, 2020311400, 2003, "De antwoorden toekomen op de schriftelijke vragen over het Nederlandse wapenexportbeleid die de heer Algra op 9 mei 2003 heeft gesteld (nr. 2020310490)".

Tweede Kamer der Staten-Generaal, documenten-en-publicaties, n.d., http://www.rijksoverheid.nl/documenten-en-publicaties/kamerstukken.html

Tweede Kamer der StaTen-.Generaal, Kamerstuk | 23-10-2007, 2007, Wapenleveranties Israel, available at:

http://www.rijksoverheid.nl/documenten-en-publicaties/kamerstukken/2007/10/23/wapenleveranties-israel.html

Tweede Kamer der Staten-.Generaal, Lijst van vragen van de vaste commissie voor Economische Zaken, 2007, Kenmerk: BEB/HP / 7144908

Tweede Kamer der Staten-.Generaal, Toezegging: Vergelijking van de controle op doorvoer van militaire goederen in eenzevental EU Lidstaten, 2007, , Kenmerk: BEB/HPG / 8013254

The Three Faces of Securitization: Political Agency, Audience and Context, available at: http://www.guillaumenicaise.com/wp-content/uploads/2014/08/Balzacq_three-faces-of-securitization.pdf

The Three Faces of Securitization: Political Agency, Audience and Context, available at: http://www.guillaumenicaise.com/wp-content/uploads/2014/08/Balzacq_three-faces-of-securitization.pdf

Trott, W., 2010, 'An analysis of civilian, military and normative power in EU foreign policy', POLIS Journal Vol. 4, University of Leeds.

Van 23 november 1999, houdende de vierentwintigste wijziging van het Uitvoerbesluit strategische goederen 1963 (bepaalde chemicaliën, test- en richtapparatuur en laserwapens, available at: C:\Users\hp\AppData\Local\Microsoft\Windows\Temporary Internet Files\Content.IE5\E3AJID43\stb-1999-516.xml

Van Evera, S., 1997, "Guide to Methods for Students of Political Science", available at: http://apps.eui.eu/Personal/Franklin/VanEvraCh1.pdf

VERSLAG VAN EEN ALGEMEEN OVERLEG Vastgesteld 9 augustus 2013, available at: https://zoek.officielebekendmakingen.nl/kst-22054-230.html

Waever, Ole, 2004, "Aberystwyth, Paris, Copenhagen New 'Schools' in Security Theory and their Origins between Core and Periphery", University of Copenhagen.

Waever, O., 2012, "Security: A Conceptual History for International Relations", Version 6

Wetgeving Vlaamse wapenhandel, 2006, "Iran en India - meten met twee maten, Besluitvorming Uruzgan - aanzetten tot een reconstructie", available at: http://vredesbeweging.nl/archief/tka06/tka0602.pdf

Wet internationale misdrijven, "Wet van 19 juni 2003, houdende regels met betrekking tot ernstige schendingen van het internationaal humanitair recht (Wet internationale misdrijven)", http://wetten.overheid.nl/BWBR0015252/geldigheidsdatum_14-12-2014#1

Whelan, C., 2012, "Networks and National Security, Ashgate Publishing Limited, England, Ashgate Publishing Company, USA.

Willardson, S., L., 2013, "Under the influence Of arms: the foreign policy causes and consequences of arms transfers", available at: http://ir.uiowa.edu/cgi/viewcontent.cgi?article=4789&context=etd

ZITTING 1960—1961, 6178, "Advies inzake het voorontwerp van wet tot regeling van de invoer, de uitvoer en de doorvoer van goederen", available at: http://resourcessgd.kb.nl/SGD/19601961/PDF/SGD 19601961 0002435.pdf

Zitting 1968-1969 - 10 042, Verlenging van de werking van krachtens de In- en uitvoerwet fStb. 7962, 295) ten aanzien van de in- en uitvoer van goederen gestelde regelen. VOORLOPIG VERSLAG, available at: http://resourcessgd.kb.nl/SGD/19681969/PDF/SGD 19681969 0003515.pdf

Zitting 1965-1966-8521, Verlenging van de werking van krachtens de In- en uitvoerwet (Stb. 1962, 295) ten aanzien van de in- en uitvoer van goederen gestelde regelen, available at: http://resourcessgd.kb.nl/SGD/19651966/PDF/SGD 19651966 0002715.pdf

Zitting 1976-1977, Nieuwe wettelijke regeling inzake het vervaardigen, verhandelen, vervoeren, voorhanden hebben, dragen enz. van wapens en munitie (Wet wapens en munitie), available at: http://resourcessgd.kb.nl/SGD/19761977/PDF/SGD 19761977 0004395.pdf

Zitting 1977-1978, Verlenging van de werking van krachtens de In-en uitvoerwet (Stb. 1962,295) ten aanzien van de in-en uitvoer van goederen gestelde regelen, available at: http://www.statengeneraaldigitaal.nl/document/tekst?id=sgd%3A19771978%3A0004310&pagina=1

Zwierlein, C., and De Graaf, B., "Security and Conspiracy in History, 16th to 21st Century", Historical Social Research, No. 143, HSR Vol. 38 (2013) 1.